#### SECURITIES AND EXCHANGE COMMISSION

### FORM 8-K

Current report filing

Filing Date: 2024-11-06 | Period of Report: 2024-10-31 SEC Accession No. 0001104659-24-114845

(HTML Version on secdatabase.com)

#### **FILER**

#### **Primis Financial Corp.**

CIK:1325670| IRS No.: 201417448 | State of Incorp.:VA | Fiscal Year End: 1231

Type: 8-K | Act: 34 | File No.: 001-33037 | Film No.: 241432175

SIC: 6022 State commercial banks

Mailing Address SUITE 900 MCLEAN VA 22102

**Business Address** 1676 INTERNATIONAL DRIVE1676 INTERNATIONAL DRIVE SUITE 900 MCLEAN VA 22102 (703) 893-7400

# UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM	8-K
------	-----

## CURRENT REPORT PURSUANT TO SECTION 13 OR 15(D) OF THE SECURITIES EXCHANGE ACT OF 1934

Date of Report (Date of earliest event reported): October 31, 2024

#### PRIMIS FINANCIAL CORP.

(Exact name of registrant as specified in its charter)

Virginia
(State or other jurisdiction of incorporation)

001-33037

20-1417448

(Commission File Number)

(IRS Employer Identification No.)

1676 International Drive, Suite 900 McLean, Virginia 22102

(Address of Principal Executive Offices) (Zip Code)

(703) 893-7400

(Registrant's telephone number, including area code)

#### Not Applicable

(Former name or former address, if changed since last report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

	COMMON STOCK	FRST	NASDAQ		
	Title of each class	Trading Symbol(s)	Name of each exchanged on which registered		
Securities registered pursuant to Section 12(b) of the Act:					
	Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))				
	Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))				
	Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)				
	Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)				

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter). Emerging growth company

	rging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. □
Item 2.01	Completion of Acquisition or Disposition of Assets.
entered in Finance D	ctober 24, 2024, Primis Bank (the wholly owned subsidiary of Primis Financial Corp.) and EverBank, N.A. ("EverBank") to a Purchase and Assumption Agreement (the "Purchase Agreement") whereby EverBank will acquire the Life Premium vivision from Primis Bank including all the associated loans, except for a subset of mostly fixed rate and rate-capped loans y Primis Bank, and the operations of the division, including its employees, for a premium of \$6.0 million (the "Transaction").
Primis Ba Between t	ctober 31, 2024, Primis Bank completed the sale to EverBank of approximately \$354 million of loans from the division with nk providing interim servicing until the transition of the business at the final closing which is expected on January 31, 2025. he first and second closings, EverBank will purchase an additional \$19 million of loans and loans generated by the division in ourse at par. After the second closing, EverBank will service the Bank's retained portfolio for the duration of the portfolio.
text of the	pregoing description of the Transaction does not purport to be complete and is qualified in its entirety by reference to the full a Purchase Agreement, a copy of which is filed as Exhibit 2.1 to this Current Report on Form 8-K and incorporated herein ce. The Transaction constituted a significant disposition for purposes of Item 2.01 of Form 8-K. Accordingly, the pro formation required by Item 9.01 is included as Exhibit 99.1 to this Current Report on Form 8-K.
Item 9.01	Financial Statements and Exhibits.
(b) Pro Fo	rma Financial Information
	ving financial information is included as Exhibit 99.1 to this Current Report on Form 8-K and is filed herewith and ed herein by reference:
Į	Unaudited Pro Forma Condensed Consolidated Balance Sheet as of September 30, 2024 Unaudited Pro Forma Condensed Consolidated Statement of Income for the year ended December 31, 2023 and the nine nonths ended September 30, 2024
(d) Exhibi	ts
2.1	Purchase and Assumption Agreement, dated October 24, 2024 by and between Primis Bank and EverBank, National Association.
<u>99.1</u>	Unaudited Pro Forma Condensed Consolidated Balance Sheet as of September 30, 2024 and Unaudited Pro Forma Condensed Consolidated Statement of Income for the year ended December 31, 2023 and the nine months ended September 30, 2024.
104	Cover Page Interactive Data File - the cover page XBRL tags are embedded within the Inline XBRL document

#### **SIGNATURES**

Pursuant to the requirements of the Securities Exchange Act of 1934, as amended, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

#### PRIMIS FINANCIAL CORP.

November 6, 2024

By: /s/ Matthew A. Switzer

Matthew Switzer Chief Financial Officer

### PURCHASE AND ASSUMPTION AGREEMENT FOR THE PRIMIS LIFE LENDING BUSINESS

#### by and between

#### EVERBANK, NATIONAL ASSOCIATION

#### and

#### **PRIMIS BANK**

#### Date October 24, 2024

#### **TABLE OF CONTENTS**

		Page
Article I I	Definitions	1
1.1	Definitions	1
1.2	Principles of Construction	6
1.3	Schedules	6
Article II	Purchase and Assumption	7
2.1	First and Second Closings	7
2.2	First Closing - Purchase and Sale of the Existing Loans	7
2.3	Purchase and Sale of the Subsequent Loans.	7
2.4	Second Closing – Purchase and Sale of the Other Acquired Assets	8
2.5	Acceptance and Assumption	9
2.6	Excluded Liabilities	9
2.7	Seller's Deliveries at each Closing	10
2.8	Buyer's Deliveries at the Closing	11
2.9	Treatment of Acquisition	12
Article III	I Purchase Price; Payment	12

3.1	Purchase Price	12
3.2	Payment	12
3.3	Preliminary Settlement Statement	12
3.4	Post-Closing Adjustments	13
3.5	Purchase Price Allocation	14
3.6	Updates to Loans Prior to the Closing Date	14
Article IV	V Covenants of Seller Prior to Closing	15
4.1	Operation of the Business	15
4.2	Books and Records	16
4.3	Representations, Warranties, Covenants and Conditions	17
4.4	Cooperation	17
4.5	Consents	17
4.6	Non-Compete	17
Article V	Covenants of Buyer Prior to Closing	17
5.1	Representations, Warranties, Covenants and Conditions	17
5.2	Cooperation	17
5.3	Consents	17
Article V	I Representations and Warranties	18
6.1	Representations and Warranties of Seller	18
6.2	Representations and Warranties of Buyer	22
Article V	II Actions Respecting Employees and Employee Benefit Plans	23
7.1	Employment of Employees; Retention	23
7.2	Terms and Conditions of Employment	24
Article V	TIII Conditions Precedent to Closing	25
8.1	Conditions to Seller's Obligations	25
8.2	Conditions to Buyer's Obligations	25

8.3	waivers of Conditions Precedent	20
Article IX	Certain Transitional Matters	27
9.1	Notices	27
9.2	Loans	27
9.3	Interest Reporting	27
9.4	Data Conversion	27
9.5	Transition Services Agreement	28
Article X	Additional Covenants	28
10.1	Further Assurances	28
10.2	Information After Closing	28
10.3	Timing of Closing	28
10.4	Termination of Intercompany Arrangements	28
10.5	Tax Matters	29
Article XI	Termination	29
11.1	Termination	29
11.2	Effect of Termination	30
Article XI	I Indemnification	30
12.1	Indemnification by Seller	30
12.2	Indemnification by Buyer	30
12.3	Third-Party Claims	30
12.4	Survival; Limitations	31
12.5	Remedies	32
Article XI	III Miscellaneous Provisions	32
13.1	Entire Agreement	32
13.2	Binding Effect; Assignment	32
13.3	Amendment and Modification	32
13.4	Waiver or Extension	32

13.5	Payment of Expenses	32
13.6	Confidentiality; Press Releases	33
13.7	Notices	33
13.8	Counterparts	34
13.9	Governing Law	34
13.10	Severability	34
13.11	No Third-Party Rights	34
13.12	Construction	34
13.13	Schedules and Exhibits	34

#### LIST OF SCHEDULES

Schedule 1.1(a)	Intellectual Property
` '	1 2
Schedule 1.3(a)	Existing Loans
Schedule 1.3(b)	Excluded Loans
Schedule 1.3(c)	Participated Loans
Schedule 1.3(d)	Secured L/C Loans
Schedule 2.4(a)	Customer List
Schedule 2.4(b)	Salesforce Technology
Schedule 2.4(d)	Assumed Contracts
Schedule 3.3	Preliminary Settlement Statement
Schedule 3.4	Final Settlement Statement
Schedule 3.5	Allocation Methodology
Schedule 6.1(g)	Employees
Schedule 6.1(j)	Consents
Schedule 6.1(m)	Broker
Schedule 6.2(g)	Broker

#### LIST OF EXHIBITS

Exhibit A	Bill of Sale and Receipt
Exhibit B	Assignment and Assumption of Loans Agreement
Exhibit C	Seller Closing Certificate
Exhibit D	Power of Attorney
Exhibit E	Transition Services Agreement
Exhibit F	Buyer Closing Certificate
Exhibit G	Forward Flow Loan Criteria
Exhibit H	Interim Servicing Agreement
Exhibit I	Servicing Agreement for Excluded Loans
Exhibit J	Intercreditor Agreement

#### PURCHASE AND ASSUMPTION AGREEMENT

This Purchase and Assumption Agreement (this "**Agreement**") is made and entered into as of October 24, 2024, by and between EverBank, National Association, a national banking association ("**Buyer**"), and Primis Bank, a Virginia state-chartered bank ("**Seller**").

#### **RECITALS:**

- A. Seller is engaged in, among other things, the operation, through Primis Life Lending, a division of Seller, of a loan origination platform throughout the United States that specializes in life premium financing and the provision of loans to ultra-high net worth individuals, trusts or related businesses that use loan proceeds to buy permanent life insurance (collectively, the "Business");
- B. Seller desires to sell to Buyer, and Buyer desires to purchase from Seller, certain properties, rights and assets of Seller used or held for use in the operation of the Business, and in connection therewith Buyer is willing to assume certain liabilities relating to the Business, all upon the terms and subject to the conditions set forth in this Agreement; and
- C. Contemporaneously with the execution of this Agreement, certain employees of and consultants to Seller (the "Acquired Employees") have entered into employment arrangements with Buyer, with effectiveness as of the Second Closing (as defined below).

#### **AGREEMENT:**

NOW THEREFORE, in consideration of the recitals, the mutual representations, covenants and agreements hereinafter set forth, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, Buyer and Seller hereby agree as follows:

#### Article I Definitions

- 1.1 <u>Definitions</u>. In addition to capitalized terms otherwise defined herein, as used in this Agreement the following capitalized terms shall have the meanings provided in this Section 1.1:
  - "Acquired Assets" shall mean all Subsequent Loans and the assets acquired under Section 2.4.
- "Acquired Customer" shall mean a Person whose Loans are acquired by Buyer hereunder. For the avoidance of doubt, no Person who is the borrower of an Excluded Loan is an "Acquired Customer" for purposes of this Agreement.
  - "Acquired Employees" shall have the meaning given such term in the Recitals.
- "Acquisition" shall mean the transactions contemplated by this Agreement, including the purchase of assets and assumption of liabilities provided for herein.
  - "Adjustment Payment" shall have the meaning given such term in Section 3.4(b).
- "Adverse Event" shall mean, with respect to any Loan, that: (a) the obligor under such Loan (i) is delinquent in the payment of principal or interest thereon for more than sixty (60) days, or (ii) has defaulted under or breached any other material provision of any agreement related to such Loan; (b) the Seller has been notified that the obligor under such Loan has filed for protection under any bankruptcy law, is insolvent or has otherwise suffered a material adverse change in its financial condition that would materially affect its ability to repay such Loan; or (c) Seller has discontinued the accrual of interest with respect to such Loan.

1

"Affiliate" of, or a Person affiliated with, a specific Person is a Person that directly or indirectly, through one or more intermediaries, controls, is controlled by, or is under common control with, the Person specified.

- "Agreement" shall have the meaning given such term in the first paragraph hereof.
- "Allocation" shall have the meaning given such term in Section 3.5.
- "Arbitrator" means a firm of independent accountants of regionally recognized standing (other than a firm that has performed services for the Seller or the Buyer or any of their Affiliates during the past three years) mutually agreeable to the Seller and the Buyer.
  - "Assumed Contracts" shall have the meaning given such term in Section 2.4(d).
  - "Assumed Liabilities" shall have the meaning given such term in Section 2.5(d).
  - "Business" shall have the meaning given such term in the Recitals.
- "Business Day" shall mean any day other than a Saturday, Sunday or a day on which banking institutions in the Commonwealth of Virginia or State of Florida are required to be closed in accordance with applicable law or regulation. Any action, notice or right that is to be taken or given or that is to be exercised or lapses on or by a given date that is not a Business Day may be taken, given or exercised and shall not lapse until the next Business Day following.
  - "Buyer" shall have the meaning given such term in the first paragraph hereof.
  - "Buyer Indemnified Party" shall have the meaning given such term in Section 12.1.
  - "Cap" shall have the meaning given such term in Section 12.4.
  - "Closing" shall mean either the First Closing or Second Closing, as applicable.
  - "Closing Date" shall mean the date on which the First Closing or Second Closing occurs, as applicable.
  - "CRA" shall mean the Community Reinvestment Act of 1977, as amended.
  - "Customer List" shall have the meaning given such term in Section 2.4(a).
  - "Employees" shall have the meaning given such term in Section 6.1(g).
  - "Estimation Date" shall have the meaning given such term in Section 3.3.
- "Excluded Assets" shall mean any asset of the Business that is not an Existing Loan or an Acquired Asset, including, but not limited to, (a) all Intellectual Property of Seller and the Business (for the avoidance of doubt, excluding the Customer Lists, facilitator lists and any rights to Salesforce technology integrations and processes used by the Business that is to be transferred to Buyer in accordance with the terms of this Agreement), medallion program stamps, signs, URLs, domain names (and associated e-mail addresses), Internet web sites, printed supplies, stationery and documents and other materials bearing any of Seller's Intellectual Property and advertising and marketing materials and similar property or rights owned by, relating to or referencing Seller or any of its Affiliates, (b) any rights to recovery under or other interest in Seller's existing bankers blanket bond, errors and omissions policy, directors' and officers' liability policy or other similar insurance policies, (c) the Excluded Loans, (d) allowance for loan losses associated with the Loans, and (e) Tax Returns and tax-related Records of Seller or any Affiliate thereof and all rights with respect to refunds, credits or similar benefits of Seller or any Affiliate thereof with respect to Taxes. The Excluded Assets shall not be sold to Buyer pursuant to this Agreement and will be retained by Seller.

- "Excluded Liabilities" shall have the meaning given such term in Section 2.6.
- "Excluded Loans" means (a) those fixed rate life premium finance loans originated by the Business, including those listed on Schedule 1.3(b), and (b) any floating rate life premium finance loan originated by the Business after the date of this Agreement which does not satisfy the criteria set forth in Exhibit G.

"Existing Loans" means those loans listed on Schedule 1.3(a) attached hereto which are floating rate life premium finance loan portfolio originated by the Business as of the First Closing, including (a) any Participated Loans for which (i) the counter-party to the participation agreement has consented to the assignment of such participation agreement to Buyer or (ii) has otherwise been terminated or bought out by Buyer, in each case prior to October 31, 2024, and (b) any Secured L/C Loan for which the underlying Letter of Credit for such Loan can also be transferred to (and drawable by) Buyer in connection with such sale on October 31, 2024.

"Final Settlement Statement" shall have the meaning given such term in Section 3.4(a).

"First Closing" shall have the meaning given such term in Section 2.1(a).

"First Closing Purchase Price" shall have the meaning given such term in Section 3.1(a).

"Forward Flow Loans" means each floating rate life premium finance loan originated by the Business after the date of this Agreement and until the Second Closing which satisfies the criteria set forth in Exhibit G.

"Fundamental Representations" means, with respect to Seller, the representations and warranties set forth in clauses (a) through (d) and (m) of Section 6.1 and with respect to Buyer, the representations and warranties set forth in clauses (a) through (c) and (g) of Section 6.2.

"GAAP" shall mean generally accepted accounting principles in effect in the United States, applied on a consistent basis.

"Indemnified Party" shall have the meaning given such term in Section 12.3.

"Indemnifying Party" shall have the meaning given such term in <u>Section 12.3</u>.

"Intellectual Property" shall mean all trademarks, trade names, service marks, patents, copyrights, logos and other intellectual property (including trade secrets, know-how, inventions, discoveries, methods, processes and improvements, whether or not patentable), whether registered, the subject of an application for registration, or unregistered, of the Business that are owned by Seller or any of its Affiliates or licensed by Seller or any of its Affiliates from a third party, including the name "Primis Life Lending" or any derivative thereof and as set forth on Schedule 1.1(a).

"IRS" shall mean the Internal Revenue Service.

3

"Lien" means, with respect to any property, equity interest or other asset, deed of trust, lien, encumbrance, hypothecation, lease, sublease, occupancy agreement, pledge, security interest, right of way, covenant, condition, use restriction, easement, encroachment, option or conditional sale agreement, in each case, in respect of such property, equity interest or other asset.

"Legal Requirements" shall mean any federal, state, local, municipal, foreign, international, multinational or other order, constitution, law, ordinance, regulation, rule, policy statement, directive, statute or treaty.

"Letters of Credit" shall mean the letters of credit between the Obligor and a financial institution wherein the Seller is the named beneficiary.

"Loans" means (a) the Existing Loans, (b) the Subsequent Loans, and (c) all loan agreements, notes, security or collateral agreements, financing statements, Liens, guarantees, participation agreements, and other documents relating thereto and all modifications, waivers and consents relating to any of the foregoing (collectively, "Loan Agreements"). For the avoidance of doubt, the Excluded Loans are not "Loans" for purposes of this Agreement.

"Loan Agreements" shall have the meaning given such term in the definition of Loans.

"Loan File" shall mean the electronic data with respect to the Loans that Seller has made available to Buyer prior to the date hereof.

- "Losses" shall have the meaning given such term in Section 12.1.
- "Objection Deadline" shall have the meaning given such term in Section 3.4(a).
- "Obligor" shall have the meaning given such term in Section 6.1(f)(iii).
- "Participated Loan" shall mean Loans that in whole or in part are subject to any participation agreement with any third party. The Participated Loans are listed on Schedule 1.3(c).
  - "**Permits**" shall have the meaning given such term in Section 6.1(1).
- "Permitted Exceptions" shall means (a) Liens for Taxes, assessments or other governmental charges, in each case, not yet delinquent or the amount or validity of which is being contested in good faith by (if then appropriate) appropriate proceedings and for which appropriate reserves have been established in accordance with GAAP, (b) mechanics', carriers', workers', repairers' and similar Liens arising or incurred in the ordinary course of business, (c) Liens created by Buyer or its successors and assigns, (d) Liens disclosed in the financial statements or Disclosure Schedules delivered to Buyer in connection with this Agreement, (e) Liens (other than monetary Liens) incurred in the ordinary course of business since the date set forth on such financial statements referenced in the preceding clause, (f) non-exclusive licenses to Intellectual Property granted in the ordinary course of business, and (g) statutory or contractual Liens of lessors or Liens on the lessor's or prior lessor's interest.
- "Person" shall mean any individual, sole proprietorship, partnership, joint venture, trust, unincorporated organization, association, corporation, limited liability company, institution, public benefit corporation, other entity or government (whether federal, state, county, city, municipal, local, foreign, or otherwise, including any instrumentality, division, agency, body or department thereof).
- "Pre-Closing Tax Period" shall mean, with respect to any Existing Loans (or other assets acquired in the First Closing), Acquired Asset or portion of the Business, any Tax period ending on or before the Closing in respect of such Existing Loans (or other assets acquired in the First Closing), Acquired Asset or portion of the Business or, in the case of any Straddle Period, the portion of such period ending on the date of such Closing.

- "Preliminary Settlement Statement" shall have the meaning given such term in Section 3.3.
- "**Premium**" means \$6,000,000.
- "Prior Employment Arrangements" shall have the meaning given such term in Section 7.1(a).
- "Purchase Price" shall have the meaning given such terms in Section 3.1(b).
- "Records" shall have the meaning given such term in Section 2.4(c).
- "Schedules" shall have the meaning given such term in Section 1.3.
- "Second Closing" shall have the meaning given such term in Section 2.1(b).
- "Secured L/C Loan" shall mean Loans that are secured in whole or in part by a Letter of Credit for which Seller is a beneficiary. The Secured L/C Loans are listed on Schedule 1.3(d).
  - "Seller" shall have the meaning given such term in the first paragraph hereof.
  - "Seller Indemnified Party" shall have the meaning given such in Section 12.2.

"Straddle Period" shall mean with respect to any Existing Loans (or other assets acquired in the First Closing), Acquired Asset or portion of the Business, any Tax period that includes (but does not end on) the applicable Closing Date in respect of such Existing Loans (or other assets acquired in the First Closing), Acquired Asset or portion of the Business.

"Subsequent Loans" shall have the meaning given such in Section 2.3(a).

"Subsequent Loan Purchase Date" shall have the meaning given such term in <u>Section 2.3</u>.

"Subsequent Loan Purchase Price" shall have the meaning given such term in Section 3.1(b).

"Tax" shall mean all forms of taxation imposed by any taxing authority, including all national, federal, state or local taxation (including income, value added, occupation, real and personal property, escheat or unclaimed property, social security (or similar, including FICA), gross receipts, sales, use, ad valorem, franchise, profits, license, withholding, payroll, employment, unemployment, disability, transfer, registration, excise, severance, occupation, premium or windfall profit taxes, environmental, stamp, duty, customs and other import or export duties, capital stock, alternative or add-on minimum, estimated and other taxes), charges, fees, levies or other assessments, in each case, in the nature of a tax, together with any interest, penalties, and additions to tax, whether disputed or not.

"Tax Return" shall mean any report, return, statement, declaration, notice, claim for refund, information return, certificate or other document filed or required to be filed with any taxing authority, including any schedule or attachment thereto, and including any amendment thereof, in connection with the determination, assessment, collection or payment of any Tax.

5

"Termination Date" shall mean March 31, 2025.

- 1.2 <u>Principles of Construction</u>. In this Agreement, unless otherwise stated or the context otherwise requires, the following usages apply:
- (a) Unless otherwise provided herein, actions permitted but not required under this Agreement may be taken at any time, and from time to time, in the actor's sole discretion;
- (b) In computing periods from a specified date to a later specified date, the words "from" and "commencing on" (and the like) mean "from and including," and the words "to," "until" and "ending on" (and the like) mean "to, and including";
- (c) Headings are inserted for convenience of reference only and are not a part of, nor shall they affect any construction or interpretation of, this Agreement;
  - (d) Unless otherwise specified, indications of time of day mean the time of day in McLean, Virginia;
- (e) All references to Articles, Sections, Schedules and Exhibits are to Articles, Sections, Schedules and Exhibits in or to this Agreement;
- (f) References to a statute shall refer to the statute and any successor statute, and to all regulations promulgated under or implementing the statute or successor statute, as in effect at the relevant time;
- (g) References to a governmental or quasi-governmental agency, authority or instrumentality shall also refer to a regulatory body that succeeds to the functions of the agency, authority or instrumentality;
  - (h) "Including" shall mean "including, but not limited to";
- (i) All words used in this Agreement in the singular number shall extend to and include the plural, all words in the plural number shall extend to and include the singular and all words in any gender shall extend to and include all genders;

- (j) Any reference to a document or set of documents in this Agreement, and the rights and obligations of the parties under any such documents, shall mean such document or documents as amended from time to time, and any and all modifications, extensions, renewals, substitutions or replacements thereof;
- (k) "except as previously disclosed" means such disclosure as communicated by Seller's counsel to Buyer's counsel prior to the date hereof; and
  - (l) All references to dollars (\$) shall mean United States currency.
- 1.3 Schedules. The schedules of Seller referred to in this Agreement (the "Schedules") shall consist of the agreements and other documentation described and referred to in this Agreement with respect to Seller, which Schedules were delivered by Seller to Buyer before the date of this Agreement. Any item or matter disclosed on any Schedule shall be deemed to be disclosed for all purposes on all other Schedules, to the extent that, it should have been disclosed on such other Schedule, to the extent that sufficient details are set forth so that the purpose for which disclosure is made is reasonably clear. In the event of any inconsistency between the statements in the body of this Agreement and those in the Schedules (other than an exception expressly set forth as such in the Schedules), the statements in the body of this Agreement will control.

#### Article II Purchase and Assumption

#### 2.1 First and Second Closings.

- (a) The acquisition of the Existing Loans shall be consummated and closed (the "**First Closing**") through the mail (including overnight courier, facsimile or e-mail) or in person at such location as shall be mutually agreed upon by Buyer and Seller, at 8:00 a.m. ET on October 31, 2024 (or on such other date or time as mutually agreed upon by Buyer and Seller), subject to the conditions set forth in <u>Article VIII</u> having been satisfied or waived as provided herein.
- (b) The acquisition of the Acquired Assets (as defined below) shall be consummated and closed (the "Second Closing") through the mail (including overnight courier, facsimile or e-mail) or in person at such location as shall be mutually agreed upon by Buyer and Seller, at 8:00 a.m. ET on January 31, 2025 (or on such other date or time as mutually agreed upon by Buyer and Seller), subject to the conditions set forth in Article VIII having been satisfied or waived as provided herein.
- 2.2 <u>First Closing Purchase and Sale of the Existing Loans</u>. Pursuant to the terms and conditions set forth herein and subject to the exceptions, if any, set forth herein, at the First Closing, Seller hereby agrees to sell, convey, assign, transfer and deliver to Buyer free and clear of all Liens and encumbrances (except for Permitted Exceptions) and Buyer hereby agrees to purchase from Seller, for the price hereinafter set forth, all of Seller's right, title and interest in:
  - (a) the Existing Loans;
  - (b) the Loan Agreements with respect to the Existing Loans and all other Records related thereto; and
- (c) the Liens, collateral, security interests, guarantees, financing statements and other rights and interests associated with the Existing Loans and/or Loan Agreements, but not including any of Seller's allowance for loan and lease losses associated with such Loans.
- 2.3 <u>Purchase and Sale of the Subsequent Loans.</u> Pursuant to the terms and conditions set forth herein and subject to the exceptions, if any, set forth herein, on the 15<sup>th</sup> and the last day of every month from the First Closing until (and including) the Second Closing (each date, a "**Subsequent Loan Purchase Date**"), Seller hereby agrees to sell, convey, assign, transfer and deliver to Buyer free and clear of all Liens and encumbrances (except for Permitted Exceptions) and Buyer hereby agrees to purchase from Seller, for the price hereinafter set forth, all of Seller's right, title and interest in:

- (a) (A) the Forward Flow Loans originated between the immediately preceding Subsequent Loan Purchase Date (or the date of the First Closing for the first purchase of Forward Flow Loans) and the applicable Subsequent Loan Purchase Date, (B) any Participated Loans for which (i) the counter-party to the participation agreement has consented to the assignment of such participation agreement to Buyer or (ii) has otherwise been terminated or bought out by Buyer, in each case prior to such Subsequent Loan Purchase Date and (C) any Secured L/C Loan for which the underlying Letter of Credit for such Loan can also be transferred to (and drawable by) Buyer in connection with such sale (collectively, the "Subsequent Loans");
  - (b) the Loan Agreements with respect to such Subsequent Loans and all other Records related thereto; and
- (c) the Liens, collateral, security interests, guarantees, financing statements and other rights and interests associated with such Subsequent Loans and/or Loan Agreements, but not including any of Seller's allowance for loan and lease losses associated with such Loans.
- 2.4 <u>Second Closing Purchase and Sale of the Other Acquired Assets</u>. Pursuant to the terms and conditions set forth herein and subject to the exceptions, if any, set forth herein, at the Second Closing, Seller hereby agrees to sell, convey, assign, transfer and deliver to Buyer free and clear of all Liens and encumbrances (except for Permitted Exceptions) and Buyer hereby agrees to purchase from Seller, for the price hereinafter set forth, all assets (other than the Existing Loans and Subsequent Loans) to the extent used in or relating to the Business conducted by Seller through Primis Life Lending other than the Excluded Assets, including the following:
- (a) <u>Customer List</u>. The list of the customers and facilitators (and past or prospective customers and facilitators) of the Business in Salesforce as set forth on Schedule 2.4(a) (the "Customer List").
- (b) <u>Technology</u>. The Salesforce technology integrations and processes used by the Business as set forth on <u>Schedule 2.4(b)</u>.
- (c) Records. All records, documents, warranties, manuals and instructions related to the Acquired Assets transferred and the liabilities described in Section 2.5 as may exist and are available to and maintained by Seller in connection with the Business, any and all documents and records in Seller's possession related to the Loans, the Loan Agreements, the transaction history of the Loans, the Letters of Credit associated with the Loans and the Assumed Contracts, including (1) credit reports, verifications (including employment verifications), underwriting or approval memoranda, financial statements of borrowers and guarantors, and independently prepared financial statements, (2) any correspondence with customers, facilitators or insurance brokers or carriers (whether in writing or in email), including management and/or resolution of any customer complaints or issues and (3) any files with respect to know your customer, anti-money laundering, sanctions or Bank Secrecy Act compliance (all such items described in this Section 2.4(c), collectively, the "Records"); and
- (d) Other Contracts and Agreements. All of Seller's right, title and interest in and to the agreements related to the Business (including all amendments, modifications, waivers, consents or approvals thereupon) as listed on Schedule 2.4(d) attached hereto (collectively, the "Assumed Contracts").

- 2.5 <u>Acceptance and Assumption</u>. Subject to the terms and conditions of this Agreement, on and as of the First Closing (only with respect to clause (a) below), each Subsequent Loan Purchase Date (only with respect to clause (b) below), or Second Closing (only with respect to clauses (c) and (d) below), Buyer shall assume, discharge and perform when lawfully due, the following obligations debts and liabilities of Seller related to the Business (in each case other than the Excluded Liabilities):
- (a) <u>Existing Loans</u>. Receive and accept all of the Existing Loans assigned, transferred, conveyed and delivered to Buyer by Seller at the First Closing pursuant to this Agreement, and assume all liabilities and obligations associated with such Existing Loans only to the extent such liabilities and obligations arise out of or relate to facts, circumstances, actions, omissions and/or events occurring after the First Closing.

- (b) <u>Subsequent Loans</u>. Receive and accept the Subsequent Loans assigned, transferred, conveyed and delivered to Buyer by Seller at each Subsequent Loan Purchase Date pursuant to this Agreement, and assume all liabilities and obligations associated with such Subsequent Loans only to the extent such liabilities and obligations arise out of or relate to facts, circumstances, actions, omissions and/or events occurring after the applicable Subsequent Loan Purchase Date.
- (c) <u>Acquired Assets</u>. Receive and accept all of the Acquired Assets assigned, transferred, conveyed and delivered to Buyer by Seller at the Second Closing pursuant to this Agreement, and assume all liabilities and obligations associated with such Acquired Assets or the Business, in each case only to the extent such liabilities and obligations arise out of or relate to facts, circumstances, actions, omissions and/or events occurring after the Second Closing.
- (d) Other Liabilities. Fully and timely perform and discharge, as the same may be or become due, Seller's obligations, duties and liabilities to the extent arising after the Second Closing under the Assumed Contracts (the liabilities expressly assumed by Buyer pursuant to this Section 2.5 (except to the extent they constitute an Excluded Liability), the "Assumed Liabilities").
- 2.6 <u>Excluded Liabilities</u>. Other than the liabilities expressly assumed by Buyer as described in Section 2.5 of this Agreement, Seller shall continue to be responsible for all known and unknown liabilities and obligations of the Business (or any other business or operations of Seller), the Acquired Assets, the Excluded Assets or the Excluded Liabilities or any of Seller's costs and expenses incurred in connection with the transactions contemplated by this Agreement (the "Excluded Liabilities"), which Excluded Liabilities include:
- (a) all liabilities or obligations relating to or arising out of the Existing Loans, to the extent such Liabilities arise out of or relate to facts, circumstances, actions, omissions and/or events (including soliciting, originating, administering or servicing any Existing Loan) occurring before the First Closing;
- (b) all liabilities or obligations relating to or arising out of the Subsequent Loans, to the extent such Liabilities arise out of or relate to facts, circumstances, actions, omissions and/or events (including soliciting, originating, administering or servicing any Subsequent Loan) occurring before the applicable Subsequent Loan Purchase Date for such Subsequent Loans;

- (c) all liabilities or obligations relating to or arising out of (i) the conduct of the Business, or (ii) any Acquired Asset, to the extent such Liabilities arise out of or relate to facts, circumstances, actions, omissions and/or events occurring before the Second Closing;
  - (d) all liabilities or obligations relating to or arising out of any Excluded Asset or any Excluded Liability;
- (e) all liabilities or obligations relating to or arising out of employment, consulting status, compensation (including any accrued but unpaid amounts or entitlements), or termination of employment or consulting role of any (A) Acquired Employee to the extent arising out of or relating to facts, circumstances, actions, omissions and/or events occurring before or at the Second Closing and (B) any current, former or prospective employees or service providers of the Business who are not Acquired Employees, whether arising prior to, as a result of, or after the Second Closing; and
- (f) all liabilities and obligations (A) for Taxes of Seller and any of its Affiliates, whether by reason of Treasury Regulation Section 1.1502-6 or otherwise, including any liability relating to, pertaining to, or arising out of the Existing Loans (or other assets acquired in the First Closing), the Business or the Acquired Assets with respect to income and other Taxes, including, but not limited to, any Taxes arising in connection with the consummation of the transaction contemplated hereby and (B) for Taxes with respect to the Existing Loans (or other assets acquired in the First Closing), Acquired Assets or the Business for any applicable Pre-Closing Tax Period.
- 2.7 <u>Seller's Deliveries at each Closing</u>. At each Closing (except as otherwise indicated below), in addition to the payments required hereunder, Seller shall deliver or cause to be delivered the following items to Buyer:
- (a) A Bill of Sale and Receipt for the Acquired Assets in the form of Exhibit A hereto, signed by a duly authorized officer of Seller (this deliverable is only required for the Second Closing).

- (b) An Assignment and Assumption of Loans Agreement in the form of Exhibit B hereto, signed by a duly authorized officer of Seller.
- (c) A Closing Certificate dated the Closing Date in the form of Exhibit C hereto, signed by a duly authorized officer of Seller.
  - (d) A validly executed IRS Form W-9 from Seller.
  - (e) A Power of Attorney in a form of Exhibit D hereto, signed by a duly authorized officer of Seller.
- (f) (i) The Acquired Assets that are capable of physical delivery (this deliverable is only required for the Second Closing) and (ii) all Records relating to Existing Loans or Acquired Assets and the liabilities described in <u>Section 2.5</u>.

- (g) Fully executed copies of the consents listed on <u>Schedule 6.1(j)</u> (this deliverable is only required for the Second Closing).
- (h) The Interim Services Agreement in the form of <u>Exhibit H</u> hereto, signed by a duly authorized officer of Seller (this deliverable is only required for the First Closing).
- (i) The Services Agreement for Excluded Loans in the form of Exhibit I hereto, signed by a duly authorized officer of Seller (this deliverable is only required for the Second Closing).
- (j) The Intercreditor Agreement in the form of <u>Exhibit J</u> hereto, signed by a duly authorized officer of Seller with respect to the Loans being transferred at the applicable Closing.
- (k) The Transition Services Agreement in the form of <u>Exhibit E</u> hereto, signed by a duly authorized officer of Seller (this deliverable is only required for the Second Closing).
- (l) Such other documents as Buyer reasonably determines are necessary to consummate the Acquisition in accordance with the terms and conditions of this Agreement.
- 2.8 <u>Buyer's Deliveries at the Closing</u>. At each Closing (except as otherwise indicated below), in addition to the payments required hereunder, Buyer shall deliver or cause to be delivered the following items to Seller:
- (a) A Bill of Sale and Receipt for the Acquired Assets in the form of Exhibit A hereto, signed by a duly authorized officer of Buyer (this deliverable is only required for the Second Closing).
- (b) An Assignment and Assumption of Loans Agreement in the form of Exhibit B hereto, signed by a duly authorized officer of Buyer.
  - (c) A Power of Attorney in the form of Exhibit D hereto, signed by a duly authorized officer of Buyer.
- (d) The Transition Services Agreement in the form of <u>Exhibit E</u> hereto, signed by a duly authorized officer of Buyer (this deliverable is only required for the Second Closing).
- (e) A Closing Certificate dated the Closing Date in the form of <u>Exhibit F</u> hereto, signed by a duly authorized officer of Buyer.
- (f) The Interim Services Agreement in the form of <u>Exhibit H</u> hereto, signed by a duly authorized officer of Buyer (this deliverable is only required for the First Closing).
- (g) The Services Agreement for Excluded Loans in the form of Exhibit I hereto, signed by a duly authorized officer of Buyer (this deliverable is only required for the Second Closing).

(h)	The Intercreditor Agreement in the form of Exhibit J hereto, signed by a duly authorized officer of Buyer
with respect to the Loans b	being transferred at the applicable Closing.

- (i) Such other documents as Seller reasonably determines are necessary to consummate the Acquisition in accordance with the terms and conditions of this Agreement.
- 2.9 <u>Treatment of Acquisition</u>. The parties intend and agree that the Acquisition and transactions contemplated by this Agreement are purchases and sales, and accordingly will treat the sale of the Existing Loans and the Acquired Assets by Seller and a purchase by Buyer pursuant to this Agreement as a sale and purchase, respectively, for legal, regulatory, accounting and tax purposes. The parties will mark their respective business records, tax returns and financial statements to reflect the transfers as sales by Seller to Buyer and purchases by Buyer from Seller. It is further not the intention of the Seller or Buyer that the conveyance of any assets hereunder is to be deemed to be a financing or pledge by Seller.

#### Article III Purchase Price; Payment

#### 3.1 Purchase Price.

- (a) At the First Closing, Buyer shall pay to Seller, in the manner specified in <u>Section 3.2</u>, funds equal to the sum of (the "**First Closing Purchase Price**"): (i) an amount equal to the aggregate book value of the Existing Loans (including accrued interest receivable) as of 12:01 am ET on the day of the First Closing; *plus* (ii) the Premium.
- (b) At each Subsequent Loan Purchase Date, Buyer shall pay to Seller, in the manner specified in <u>Section 3.2</u>, funds equal to the aggregate book value of Subsequent Loans (including accrued interest receivable) as of 12:01 am ET on the day of such Subsequent Loan Purchase Date (in the aggregate, the "Subsequent Loan Purchase Price," and together with the First Closing Purchase Price, the "Purchase Price").
- 3.2 <u>Payment</u>. Each transfer of funds under this Agreement shall be made by wire transfer of immediately available funds in accordance with the wire instructions to be provided by the party to receive such payment.
- 3.3 <u>Preliminary Settlement Statement.</u> To facilitate each Closing and/or the periodic Subsequent Loan purchases, as applicable, the parties agree that the payment made on the applicable Closing Date or the Subsequent Loan Purchase Date pursuant to Section 3.1 shall be computed using the methodology set forth in Section 3.1 but shall be based upon the aggregate principal amount of all Loans (including accrued but unpaid interest and late charges), as of the close of business on that day which is no earlier than five (5) Business Days preceding the applicable Closing Date or the Subsequent Loan Purchase Date (the "Estimation Date"). Not less than three (3) Business Days prior to the applicable Closing Date or the Subsequent Loan Purchase Date, Seller shall deliver to Buyer a preliminary settlement statement in substantially the form set forth on Schedule 3.3 (the "Preliminary Settlement Statement") together with all backup schedules and information. Buyer shall have the opportunity to review such statement and information and not less than one (1) day prior to the applicable Closing Date or the Subsequent Loan Purchase Date, Buyer shall notify Seller of any objections Buyer has to the Preliminary Settlement Statement. Prior to the applicable Closing Date or the Subsequent Loan Purchase Date, Seller and Buyer shall agree upon the information and calculations set forth on the Preliminary Settlement Statement. The Preliminary Settlement Statement shall be the basis of the payment to be made at the Closing and/or the Subsequent Loan Purchase Date, as applicable, pursuant to Section 3.1.

12

#### 3.4 <u>Post-Closing Adjustments</u>.

- (a) Within fifteen (15) Business Days after each Closing or Subsequent Loan Purchase Date, Seller shall deliver to Buyer a final settlement statement with respect to such Closing or Subsequent Loans purchase in substantially the form set forth on Schedule 3.4 (the "Final Settlement Statement"), that shall include appropriate post-closing adjustments based upon actual Loans sold and transferred to Buyer as of such Closing or Subsequent Loan Purchase Date. Buyer shall notify Seller of any disputes with respect to the Final Settlement Statement within fifteen (15) Business Days of Buyer's receipt thereof (the "Objection Deadline"). The Buyer shall afford the Seller and its accountants and attorneys reasonable access and opportunity to review the relevant supporting documentation and Records used by the Buyer in disputing the Final Settlement Statement.
- (b) The parties shall make an appropriate post-closing adjustment payment to reflect the difference, if any, between the amount of Loans calculated pursuant to the Final Settlement Statement and the amount of Loans calculated and paid pursuant to the past Preliminary Settlement Statement(s) (the "Adjustment Payment"). Interest on the amount of the Adjustment Payment for the period from the time of the applicable Closing or the Subsequent Loan Purchase Date to the date of the payment of the Adjustment Payment shall be due to the party receiving the Adjustment Payment. Interest shall be calculated at a rate equal to the average of the high and low bids for Federal Funds as reported in *The Wall Street Journal*, on the applicable Closing or the Subsequent Loan Purchase Date or, if none, on the latest date immediately prior to the applicable Closing or the Subsequent Loan Purchase Date on which such bids were reported in *The Wall Street Journal*.
- (c) If the Buyer objects to the Final Settlement Statement by the Objection Deadline, Buyer and Seller shall work in good faith to resolve any objections but, if the Buyer and Seller are unable to agree on the amount of the Adjustment Payment within thirty (30) days after Buyer timely submits its objection notice, either the Buyer or the Seller may submit the matter to an Arbitrator, which shall determine all disputed portions of the Final Settlement Statement in accordance with the terms and conditions of this Agreement within thirty (30) days after the submission of such matter for Mediation. The Buyer and the Seller shall each pay half of the fees and expenses of the Arbitrator, except that the Arbitrator may assess the full amount of its fees and expenses against the relevant party if it determines that such party presented or objected to the Final Settlement Statement in bad faith. The Final Settlement Statement and the Adjustment Payment, as agreed upon by the Buyer and the Seller or determined by the Arbitrator under this subsection, shall be final and binding upon the parties.
- (d) The Adjustment Payment and any interest thereon shall be paid by wire transfer of immediately available funds to the account of the party receiving the payment within five (5) Business Days of the Objection Deadline (if no objection is timely made by Buyer), or, in the event that Buyer timely objects to the Final Settlement Statement, within five (5) Business Days of the date that the Final Settlement Statement and the amount of the Adjustment Payment become final and binding upon the parties in accordance with subsection (c) of this Section 3.4.

3.5 Purchase Price Allocation. Seller and Buyer agree that the Purchase Price (and any other amounts treated as consideration for U.S. federal income Tax purposes, including the Assumed Liabilities) shall be allocated among the Existing Loans (or other assets acquired in the First Closing) and Acquired Assets on the basis of an allocation (the "Allocation") to be consistent with Section 1060 of the Internal Revenue Code and the methodologies set forth in Schedule 3.5. Within thirty (30) calendar days after the finalization of each of the First Closing Purchase Price and Subsequent Loan Purchase Price pursuant to Section 3.4, Buyer shall deliver to Seller a draft Allocation consistent with Schedule 3.5 with respect to each of the First Closing and the Second Closing. If within thirty (30) days after Seller's receipt of such draft Allocation Seller has not objected in writing to such draft Allocation, it shall become final. In the event that Seller objects in writing within such 30-day period, the parties shall negotiate in good faith to resolve the dispute. If Buyer and Seller cannot agree on a mutually satisfactory Allocation within 15 days after Seller timely submits its objection notice, either Buyer or Seller may submit the matter to an Arbitrator, which shall determine all disputed portions of the Allocation in accordance with the terms and conditions of this Agreement within fifteen (15) days after the submission of such matter. Buyer and Seller shall each pay half of the fees and expenses of the Arbitrator, except that the Arbitrator may assess the full amount of its fees and expenses against the relevant party if it determines that such party presented or objected to the Allocation in bad faith. Each Allocation, as agreed upon by Buyer and Seller or determined by the Arbitrator under this subsection, shall be final and binding upon the parties. The parties will report (including with respect to the filing of Forms 8594 with the IRS) the sale and purchase for all Tax purposes in a manner consistent with each Allocation and will not, in connection with the filing of any Tax Return or similar report, make any allocation of the consideration payable hereunder that is contrary thereto unless otherwise required by applicable law. The parties will consult, and cooperate, with one another with respect to any Tax audit, controversy, or litigation relating to each Allocation and IRS Forms 8594.

- 3.6 <u>Updates to Loans Prior to the Closing Date</u>. From the date of this Agreement until the applicable Closing Date, Seller shall promptly provide to Buyer: (i) written notice of the occurrence of any Adverse Event; and (ii) such other information regarding the Loans as is reasonably requested by Buyer.
- 3.7 <u>Withholding Taxes</u>. Buyer (and any other applicable withholding agent) shall be entitled to deduct and withhold from the amounts otherwise payable pursuant to this Agreement to any Person such amounts as it is required to deduct and withhold with respect to the making of such payment under any applicable Tax law; provided, that Buyer will use commercially reasonable efforts to notify such Person of amounts otherwise payable to such Persons that it intends to deduct and withhold at least five Business Days prior to the applicable Closing and shall use commercially reasonable efforts to assist such Person in reducing or eliminating such withholding. To the extent that any such amounts are so deducted and withheld and paid to the appropriate taxing authority in accordance with applicable Tax law, such withheld amounts shall be treated for all purposes of this Agreement as having been paid to the Person in respect of which such deduction and withholding was made.

### Article IV Covenants of Seller Prior to Closing

Seller hereby covenants to Buyer that, from the date hereof until the Second Closing, it will do or cause the following to occur:

#### 4.1 Operation of the Business.

- (a) Seller will conduct the operations of the Business in the ordinary course of business as conducted as of the date hereof, except for activities related to the Acquisition. In addition, except as may be required by regulatory authorities or by this Agreement, with prior consent of Buyer, or except as previously disclosed, Seller shall:
  - (i) comply in all material respects with all applicable Legal Requirements relating to the operation of the Business:
  - (ii) retain all necessary business permits, licenses, registrations and authorizations relating to the Business;
  - (iii) use commercially reasonable efforts to preserve for Buyer the goodwill of its customers and others doing business with the Business;
  - (iv) not amend, modify, renew or cancel, any Assumed Contract, other than in the ordinary course of business or as required to consummate the Acquisition;
  - (v) not sell, pledge, transfer, dispose of, encumber (other than Permitted Encumbrances), lease or license any Acquired Assets;
  - (vi) not release, compromise or waive any material claim or material right that is part of the Acquired Assets;
  - (vii) not settle or compromise any litigation or investigation if such settlement or litigation would impose any material obligation or liability on the Acquired Assets or Buyer;
  - (viii) not make, change, revoke or modify any material Tax election, file any material amended Tax Return, surrender any right to claim a material Tax refund, consent to any extension or waiver of the limitations period applicable to any Tax Return, settle or compromise any material Tax liability or file any Tax Return relating to the Business or the Acquired Assets on a basis inconsistent with past practice (unless otherwise required by applicable Law), in each case to the extent in respect of the Business or the Acquired Assets;
  - (ix) not terminate the employment or services of any Acquired Employee, except for cause, or hire any new employee of the Business;

- (xi) not establish, adopt, enter into or amend any plan, agreement or arrangement that provides incentive compensation, bonus or commissions or other compensation or benefits for the Acquired Employees, in each case, that would individually or in the aggregate result in any increase in liability for Buyer other than a de minimis increase, except as required by applicable Law or the terms of any such plan, agreement or arrangement in existence on the date hereof and disclosed in writing to Buyer or its counsel; and
- (xii) use commercially reasonable efforts to (A) preserve the Business operations as conducted by the Seller through Primis Life Lending; (B) cooperate with and assist Buyer in assuring the orderly transition of the Business (other than the Excluded Assets or Excluded Liabilities), the Loans to Buyer from Seller (including (i) maintaining any blocks or freezes on deposit accounts at Seller that serve as collateral for any Loans and (ii) providing reasonable cooperation with respect to Buyer putting in place customary collateral documentation in favor of Buyer with respect to such accounts by the Second Closing, in each case as permitted by applicable law); and (C) maintain Seller's normal and customary practices and procedures regarding loan pricing, underwriting and recognition of charge-offs in a manner consistent with past practice with respect to the Loans of the Business.
- (b) Notwithstanding the foregoing, except as may be required by regulatory authorities or by this Agreement, Seller shall not, without the prior consent of Buyer or: (i) transfer to Seller's other facilities any of the Acquired Assets other than in the ordinary course of business; or (ii) transfer, assign, encumber or otherwise dispose of or enter into any contract, agreement or understanding, or negotiate with any party with respect to entering into a contract, agreement or understanding, to transfer, assign, encumber or otherwise dispose of any or all of the Acquired Assets or pursuant to this Agreement.

#### 4.2 Books and Records.

- (a) Between the date hereof and the Second Closing, Buyer may continue its investigation and due diligence review of the Business and the Acquired Assets and Seller shall afford Buyer and its representatives reasonable access to the Business, and any properties, equipment and Records that pertain to the Business, the Acquired Assets, or the Acquisition during the normal business hours of the Seller; *provided, however*, that Buyer shall notify Seller at least two (2) Business Days prior to the time access is requested by Buyer, and *provided, further*, that the foregoing actions do not interfere with the business operations of Seller. Nothing in this Section 4.2 shall be deemed to require Seller to breach any obligation of confidentiality or to reveal any proprietary information, trade secrets or marketing, business or strategic plans or share board of directors' minutes or materials to the extent the materials address this transaction or other potential transactions of Seller (provided that Seller will use commercially reasonable efforts to provide for alternative disclosures).
- (b) From the execution of this Agreement until the Second Closing, Seller hereby provides to Buyer a restricted use license for the underlying code used to run the Salesforce technology integrations and processes in a format reasonably agreed between the parties, in order to assist Buyer in its due diligence and integration review. From and after the Second Closing, Seller shall take all actions to transfer the underlying code used to run the Salesforce technology integrations and processes to Buyer and provide the Buyer an unrestricted license to use the foregoing.

16

4.3 <u>Representations, Warranties, Covenants and Conditions</u>. Seller shall take all reasonable actions to ensure that the representations and warranties set forth in <u>Section 6.1</u> will be true, correct and complete as of each Closing in all material respects, that all covenants of Seller set forth in this Agreement that are required to be performed at or prior to each Closing shall have been performed at or prior to such Closing as provided in this Agreement in all material respects and that all conditions to Buyer's obligation to close shall have been satisfied at or prior to each Closing.

- 4.4 <u>Cooperation</u>. Following the Closings, Seller agrees to use reasonable efforts to provide Buyer with any information or data with respect to the Loans or other Acquired Assets that was not transferred to Buyer at the applicable Closing, including in connection with any regulatory inquiries or third-party claims or actions.
- 4.5 <u>Consents.</u> Seller shall use its best efforts to obtain on or before the Second Closing any required consents to assign the Assumed Contracts to Buyer.
- 4.6 Non-Compete. For a period of one year from the Second Closing, Seller shall not and shall cause its Affiliates not to, directly or indirectly, own an equity interest in, or manage, operate or control any Person engaged in, or otherwise engage in the business of originating life premium finance loans in the United States; provided that the foregoing shall not restrict Seller or its Affiliates from (a) holding the Excluded Loans, (b) holding a security in a fiduciary, custodial or agency capacity or (c) owning 5% or less of the outstanding stock of any Person that engages in such business.

### Article V Covenants of Buyer Prior to Closing

- 8.1 Representations, Warranties, Covenants and Conditions. Buyer shall take all reasonable actions to ensure that the representations and warranties set forth in Section 6.2 will be true, correct and complete as of each Closing, that all covenants of Buyer set forth in this Agreement that are required to be performed at or prior to each Closing shall have been performed at or prior to each Closing as provided in this Agreement and that all conditions of Seller's obligations to close the Acquisition shall have been satisfied at or prior to each Closing.
- 5.2 <u>Cooperation</u>. Following the Closings, Buyer agrees to use reasonable efforts to provide Seller with any information or data with respect to the Loans or other Acquired Assets that were transferred to Buyer to the extent Seller reasonably requires such information in connection with its prior ownership of the Loans or other Acquired Assets, including in connection with any regulatory inquiries or third party claims or actions with respect to its ownership of the Loans or other Acquired Assets.
- 5.3 <u>Consents.</u> Buyer shall use its best efforts to obtain on or before the Second Closing any required consents to consummate the Acquisition and/or assume the Assumed Contracts from Seller.

17

#### Article VI Representations and Warranties

- Representations and Warranties of Seller. Except as previously disclosed, Seller represents and warrants to Buyer on the date hereof and as of each of the First Closing (provided no representations and warranties are made with respect to the Acquired Assets other than the Existing Loans as of the First Closing) and the Second Closing (or such other date(s) as specified, to the extent any representation or warranty speaks as of a specific date) as follows:
- (a) Good Standing and Power of Seller. Seller is a Virginia state-chartered bank, duly organized, validly existing and in good standing with all requisite power and authority to own its properties and to carry on its business as presently conducted.
- (b) <u>Authorization of Agreement</u>. The execution, delivery and performance of this Agreement and the Acquisition have been duly authorized by all necessary corporate action on the part of Seller, and this Agreement is a valid and binding obligation of Seller enforceable against Seller in accordance with its terms, except to the extent enforcement may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting creditors' rights generally and by general principles of equity (regardless of whether enforcement is sought in equity or at law). Seller has the requisite power and authority to enter into and perform its obligations under this Agreement.
- (c) <u>Effective Agreement</u>. The execution, delivery and performance of this Agreement by Seller and the consummation of the Acquisition (i) does not require Seller or its Affiliates to obtain any authorization, waiver, consent or approval of, or make any filing or registration with, or give any notice to, any government authority, (ii) does not conflict with, result in the breach of, constitute a violation or default under, any of the provisions of Seller's articles of incorporation or bylaws or any judgment, decree

or order, any law, rule or regulation of any government or agency thereof, or any agreement, contract or instrument to which Seller is subject, and (iii) does not conflict with, result in the breach of, constitute a violation or default under, any (x) Acquired Assets or Assumed Liabilities or (y) other contracts to which Seller is a party, which, with respect to this clause (iii) would materially adversely impact Seller's ability to perform its obligations hereunder.

(d) <u>Title to Assets</u>. Seller is the sole owner of each of the Acquired Assets (including the Loans), free and clear of any Lien, encumbrance or restriction of any kind or nature, other than the Permitted Exceptions.

#### (e) Taxes and Insurance.

(i) Seller has timely filed, or has caused to be timely filed on its behalf, all material Tax Returns in respect of or in relation to the Existing Loans (or other assets acquired in the First Closing), the Business or the Acquired Assets required to be filed by or with respect to it (taking into account any extensions of time in which to file). All such Tax Returns are true, correct and complete in all material respects. Seller has paid all income and other material Taxes in respect of or in relation to the Existing Loans (or other assets acquired in the First Closing), the Business or the Acquired Assets imposed by any taxing authority which are due and payable by Seller as of the date hereof and there are no unpaid Taxes which could result in material Liens being placed on the Existing Loans (or other assets acquired in the First Closing) or the Acquired Assets, other than unpaid Taxes on collateral securing Loans and any Permitted Exceptions.

18

- (ii) Seller has deducted, withheld, and timely paid to the appropriate taxing authority all material Taxes required to be deducted, withheld or paid with respect to or in relation to the Business or any Acquired Asset or any Existing Loan (or other assets acquired in the First Closing).
- (iii) There is no pending action, suit, proceeding, investigation, dispute, audit or claim concerning any liability for a material amount of Taxes of Seller with respect to or in relation to the Business or any Acquired Asset or any Existing Loan (or other assets acquired in the First Closing) that is currently pending, or has been claimed, threatened or raised by any taxing authority in writing. No deficiency for a material amount of Taxes has been asserted or assessed by a taxing authority in writing with respect to the Business or the Acquired Assets or the Existing Loans (or other assets acquired in the First Closing) that has not been satisfied by payment or otherwise settled or withdrawn.
- (iv) No waivers or extensions of statutes of limitations or time in which to file Tax Returns (other than waivers or extensions no longer in force or automatic extensions) have been granted for any claim for, or period for the collection or assessment of, Taxes in respect of or in relation to the Business or any Acquired Asset or any Existing Loan (or other assets acquired in the First Closing).
- (v) Seller shall maintain in full force and effect through the Closing Date its present insurance coverage as it relates to the Acquired Assets and the Existing Loans (or other assets acquired in the First Closing).

#### (f) Loans.

- (i) The Seller has full power and authority to hold the Loans, and has good and marketable title to its Loans free and clear of all Liens and encumbrances. The Seller is authorized to sell and assign its Loans to the Buyer and, upon assignment, the Buyer shall have the rights of the Seller with respect to the Loans.
- (ii) Each Loan was solicited and originated, and is and has been administered and, where applicable, serviced, and the relevant Loan files are being maintained, in all material respects in accordance with the relevant notes or other credit or security documents, the written underwriting standards of Seller and with all applicable federal, state and local laws, regulations and rules. The principal balance of each Loan as shown on the Loan File is true and correct as of the last day shown thereon. No Adverse Event has occurred in respect of any Loan. The Seller has complied in all material respects with all of its obligations under its Loan Agreements.
- (iii) Each of the Loans (i) is the legal, valid and binding obligation of the obligor, maker, co-maker, guarantor, endorser or debtor (such person referred to herein as an "Obligor") thereof or thereunder and is evidenced by legal,

valid and binding instruments executed by the respective Obligors, each of which at the time of such execution had capacity to contract, and any signature on any Loan Agreements is the true signature of the Obligor on the Loan involved, and (ii) is enforceable in accordance with its terms (except as enforceability may be limited by bankruptcy laws and other laws of similar nature relating to creditors rights and to general principles of equity).

1	$\sim$
	u

- (iv) The documentation relating to each Loan and the Loan File accurately reflects in all material respects the payment history, all of the terms and conditions and the outstanding balance of each Loan, and all receipts pertaining to each Loan from the Obligor thereof and all credits to which such Obligor is entitled.
  - (v) Prior to the date hereof, Seller has provided to Buyer access to the file of each Loan.
- Employee Matters. A list of employees and consultants associated with the Business is set forth on Schedule 6.1(g) (the "Employees"), including, as applicable, the position, corporate and functional title, date of commencement of employment or engagement, annual base salary or rate of pay, target annual cash incentive compensation opportunities, status as exempt or non-exempt, identification number, status as full or part-time, status as active or on leave, if on leave, the date leave commenced and geographic location of each such Employee. Seller has complied in all material respects with all Legal Requirements relating to labor and the employment or engagement of the Employees, including provisions relating to wages, hours, collective bargaining and the payment of social security or other Taxes, and worker's compensation or other insurance premiums with respect to the Employees. Except as set forth on Schedule 6.1(g), there are no written employment or consulting contracts, arrangements or similar agreements between Seller and any of the Employees not terminable on thirty (30) days' notice after the Closing Date of the Second Closing. Seller is not a party to any contract or arrangement with any union relating to the Business and Seller is not aware of any pending organizational efforts regarding the Business nor has Seller experienced any labor strike, slowdown or work stoppage, lockout, unfair labor practice charges, grievances, arbitration or other material labor controversy relating to the Business within the past two (2) years.
- (h) Third-Party Claims. There are no claims, actions, suits, hearings, investigations or proceedings of any kind in any court or by or before any governmental authority or arbitration board or tribunal, that are pending or, to Seller's knowledge, threatened against or affecting (x) the Business, the Acquired Assets or the Assumed Liabilities or (y) otherwise involving Seller, that with respect to this clause (y), if determined adversely to Seller, would materially and adversely impact Seller's ability to perform its obligations hereunder or prohibit consummation of the Acquisition. To Seller's knowledge, there is no decree, judgment, or order of any kind in existence against, affecting or restraining (x) the Business, the Acquired Assets or the Assumed Liabilities or (y) otherwise involving Seller, or any of its officers, employees, or directors that with respect to this clause (y), would materially and adversely impact Seller's ability to perform its obligations hereunder.
- (i) <u>Governmental Notices</u>. Except as disclosed, Seller has no reason to believe that there are any regulatory restrictions regarding its ability to complete the transaction contemplated hereby in the manner herein contemplated and has received no written notice or written communication from any state or federal banking regulatory agency or authority indicating that such agency or authority would, and Seller has no reason to believe any such agency or authority would, object to the consummation by Seller of the Acquisition. The most recent CRA rating of Seller was not less than "satisfactory."

20

(j) Assumed Contracts. The Assumed Contracts constitute the legal, valid and binding obligations of Seller and the other parties thereto, enforceable in accordance with their terms (except as enforceability may be limited by bankruptcy laws and other laws of a similar nature relating to creditors rights). Seller is not in material default or breach under any Assumed Contracts and, to the knowledge of Seller, no other party to any of the Assumed Contracts (x) is in material default or breach thereunder or (y) has alleged that Seller is in material default or breach of any Assumed Contract or has otherwise notified Seller of its intent to terminate, not renew or increase pricing with respect to any Assumed Contract. Except as set forth on Schedule 6.1(j), each of the Assumed Contracts may be assigned to Buyer by Seller without the approval or consent of any other Person. Seller has delivered to Buyer true and correct copies of each of the Assumed Contracts and all attachments and addenda thereto.

- (k) <u>Collateral</u>. With respect to each Loan that is secured by collateral, each such Loan has the security position set forth in the Loan File as of the date hereof. No collateral has been released from the Lien granted to Seller or subordinated to a claim of a third party (or any other claim against the borrower held by Seller) unless approved by Seller and documented in the Records and Loan File. As of the date hereof, Seller has not issued any letters of credit included in, or in respect of, any Loans.
- (l) Compliance with Laws. The Business has been for the past three (3) years and is being conducted in material compliance with all applicable laws, rules, and regulations of all authorities, including, but not limited to, compliance with anti-money laundering, sanctions and anti-corruption laws and regulations, The Currency and Foreign Transaction Reporting Act of 1970 and all other regulations pertaining to the receipt of customer information required by state and federal law concerning taxpayer identification numbers and social security numbers. The Seller has all permits, licenses, registrations and authorizations ("Permits") necessary for the conduct and operations of the Business as currently conducted in all material respects, and the Business has been for the past two (2) years and is being conducted in material compliance with all such Permits and the Seller has not received any notification of breach, violation, suspension or cancellation of any Permit.
- (m) No Broker. Other than as described on Schedule 6.1(m), no broker, finder or any other party or agent performing similar functions has been retained by Seller or its Affiliates or is entitled to be paid based on any arrangements, agreements or understandings made by Seller or its Affiliates in connection with the Acquisition. Any fees or expenses of any such Person (including any fees relating to the consummation of the transactions contemplated hereby) shall be borne solely by Seller.
- (n) <u>Customer List. Schedule 2.4(a)</u> contains a complete and correct Customer List as of the date of this Agreement. All of the customer accounts set forth in the Customer List are active and in full force and effect and to Seller's knowledge, no fact, issue or circumstance exists which would materially and adversely impact any customer on the Customer List. Seller has not received any notice that any of the account holders on the Customer List has terminated its relationship with Seller or the Business or intends to terminate its relationship with Seller of the Business.

- (o) Records. The Records contain, in all material respects, complete copies of all the Loan Agreements and other books and records related to the Business. The Records (i) have been maintained in all material respects in accordance with Seller's policies and procedures applied on a consistent basis, (ii) are accurate and complete in all material respects and (iii) provide an accurate and complete accounting in all material respects for each Loan and have been maintained in accordance with the requirements of each Loan in all material respects.
- (p) <u>Sufficiency of Assets</u>. Aside from the Employees, the Acquired Assets and Assumed Liabilities constitute sufficient assets, rights and services necessary to conduct the Business in all material respects as currently conducted. The Seller is the only entity through which the Business is conducted.
- (q) <u>IP</u>. The conduct or operations of the Business is not infringing, misappropriating or otherwise violating any third party's Intellectual Property in any material respects and Seller has not received any material claims or allegations with respect thereto.
- (r) <u>Non-Competes/Non-Solicits</u>. The Business or Seller is not subject to any agreement or understanding with any Person from carrying on the Business in any geographic location (including soliciting new customers) or from expanding the Business or from soliciting or hiring employees for the Business.
- 6.2 <u>Representations and Warranties of Buyer</u>. Buyer represents and warrants to Seller on the date hereof and as of each Closing Date (or such other date(s) as specified, to the extent any representation and warranty speaks as of a specific date) as follows:
- (a) <u>Good Standing and Power of Buyer</u>. Buyer is a national banking association, duly organized, validly existing and in good standing with all requisite power and authority to own its properties and to carry on its business as presently conducted.
- (b) <u>Authorization of Agreement</u>. The execution, delivery and performance of this Agreement and the Acquisition have been duly authorized by all necessary corporate action on the part of Buyer, and this Agreement is a valid and binding obligation of Buyer enforceable against Buyer in accordance with its terms, except to the extent enforcement may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting creditors' rights generally and by general principles of equity

(regardless of whether enforcement is sought in equity or at law). Buyer has the requisite power and authority to enter into and perform its obligations under this Agreement.

(c) <u>Effective Agreement</u>. The execution, delivery and performance of this Agreement by Buyer and the consummation of the Acquisition (i) does not require Buyer or its Affiliates to obtain any authorization, waiver, consent or approval of, or make any filing or registration with, or give any notice to, any Government Authority, and (ii) does not conflict with, result in the breach of, constitute a violation or default under, any of the provisions of Buyer's charter or bylaws, any judgment, decree or order, any law, rule or regulation of any government or agency thereof, or any material agreement, material contract or material instrument to which Buyer is subject, which, with respect to clause (ii), where such conflict, breach, violation, default, acceleration or Lien would have a material adverse effect on Buyer's ability to perform its obligations hereunder.

22

- (d) <u>Compliance with Law.</u> The business and operations of Buyer have been and are being conducted in material compliance with all laws, rules, and regulations of all authorities, except to the extent it would not have a material adverse effect on Buyer's ability to consummate the transactions contemplated hereunder. The Buyer is in compliance with all of its regulatory capital standards as of the date hereof.
- (e) <u>Third-Party Claims</u>. There are no claims, actions, suits, hearings, investigations or proceedings of any kind in any court or by or before any governmental authority or arbitration board or tribunal that are pending or threatened against or affecting Buyer that, if determined adversely to Buyer, could: (i) reasonably be expected to have a material adverse effect on Buyer's ability to consummate the Acquisition; or (ii) prohibit or materially delay the consummation of the Acquisition. There is no decree, judgment, or order of any kind in existence, against, affecting or restraining Buyer or any of its officers, employees, or directors from taking any actions of any kind in connection with the performance of this Agreement and the Acquisition.
- (f) <u>Financial Resources</u>. Buyer has sufficient capital and cash available and, on the Closing Date, will have sufficient capital and cash available to perform its obligations under this Agreement and Buyer's ability to purchase the Acquired Assets and to assume such liabilities as provided herein, and to perform Buyer's other obligations hereunder is not contingent on raising any equal capital, obtaining specific financing thereof or obtaining the consent of any lender.
- (g) <u>No Broker</u>. Other than as described on <u>Schedule 6.2(g)</u>, no broker, finder or any other party or agent performing similar functions has been retained by Buyer or its Affiliates or is entitled to be paid based on any arrangements, agreements or understandings made by Buyer or its Affiliates in connection with the Acquisition, and no brokerage fee or other commission has been agreed to be paid by Buyer or its Affiliates on account of the Acquisition.
- (h) <u>Governmental Notices</u>. There are no regulatory restrictions regarding Buyer's ability to complete the transaction contemplated hereby in the manner herein contemplated, including by way of delay, and Buyer has received no notice or communication, in any form, from any state or federal banking regulatory agency or authority indicating that such agency or authority would, and Buyer has no reason to believe any such agency or authority would, object to, or withhold any approval or consent necessary for, the consummation by Buyer of the Acquisition. There are no pending or threatened legal or governmental proceedings against Buyer or any Affiliate of Buyer that would affect Buyer's ability to obtain the regulatory approvals, if any, required in order to consummate the Acquisition.

23

### Article VII Actions Respecting Employees and Employee Benefit Plans

#### 7.1 Employment of Employees; Retention.

(a) Between the date of this Agreement and the Closing Date of the Second Closing, Buyer shall have the opportunity to meet with in person and to interview each Acquired Employee. The agreements between Buyer and the Acquired

Employees containing an offer of employment are conditional on (i) the successful completion of Buyer's background check and other pre-hire requirements (including execution of Buyer's customary confidentiality and non-solicitation agreement), and (ii) the written termination of any applicable employment agreements and/or restrictive covenants between the applicable Acquired Employee and Seller or its Affiliates (the "**Prior Employment Arrangements**"). Buyer shall have no obligations in respect of, and Seller shall indemnify and hold harmless the Buyer and its Affiliates for, any and all severance and other payments or benefits arising under the Prior Employment Arrangements, which shall be treated for purposes of this Agreement as Excluded Liabilities. Nothing contained in this Agreement shall be construed as creating an employment agreement between Buyer and an employee of the Business.

(b) Seller shall take, or shall cause to be taken, all actions reasonably required to release, or cause to be released, each Acquired Employee, effective at or before the Second Closing, from any agreement or contract with, or other obligation to, Seller or its Affiliates to the extent such agreement, contract or obligation limits or restricts such Acquired Employee from being employed by or providing services to Buyer or any of its Affiliates, or otherwise engaging in the Business on behalf of Buyer or any of its Affiliates.

#### 7.2 <u>Terms and Conditions of Employment.</u>

- (a) General. The terms of employment for each Acquired Employee shall be determined solely by Buyer in its sole and absolute discretion. Notwithstanding the foregoing, (i) Buyer shall provide such Acquired Employees with credit for the Business Employee's period of service with Seller towards the calculation of eligibility and vesting with respect to employee benefit and welfare plans (other than under any employee stock ownership, stock award plan or other equity or equity-based plan), and (ii) each Acquired Employee shall be eligible to participate in the medical, dental and other benefit plans of Buyer, as such plans may exist, on and after the Closing Date of the Second Closing to the same extent as other employees of Buyer, and shall be provided with bank and banking privileges substantially similar to those provided to similarly situated employees of Buyer. Buyer shall use commercially reasonable efforts subject to the terms of any plan to cause each such plan to (i) provide full credit under such plans for any deductibles, co-payment and out-of-pocket expenses incurred by Business Employees and their beneficiaries during the portion of the calendar year prior to such participation as if such amounts had been paid in accordance with such plan of Seller; and (ii) waive any waiting period limitation, evidence of insurability or actively-at-work requirement which would otherwise be applicable to such employee on or after the Closing Date of the Second Closing to the extent such employee had satisfied any similar limitation or requirement under an analogous Seller plan. Nothing in this Section 7.2 or otherwise shall prohibit a Acquired Employee from transferring such Acquired Employee's 401(k) fund to an individual retirement account or other self-directed program at or after the Second Closing.
- (b) <u>Division of Obligations</u>. Except as provided herein, Seller shall pay, discharge and be responsible for: (i) all salary and wages or other compensation arising out of employment or engagement of the Employees and Acquired Employees through the Second Closing; and (ii) any employee benefits (including vacation, sick and personal days accrued but unused) arising under Seller's employee benefit plans and employee programs at any time (all of which shall be Excluded Liabilities hereunder). From and after the Second Closing, Buyer shall pay, discharge and be responsible for all salary, wages and benefits and any other compensation arising out of or relating to the employment or engagement of the Acquired Employees by Buyer from and after the Second Closing Date, including all claims incurred by participating Employees under Buyer's welfare benefits plans on or after the Second Closing Date.

24

### Article VIII Conditions Precedent to Closing

- 8.1 <u>Conditions to Seller's Obligations</u>. The obligation of Seller to consummate the Acquisition is subject to the satisfaction, or the waiver in writing by Seller to the extent permitted by applicable law, of the following conditions at or prior to the Closing:
- Representations and Warranties. The representations and warranties of Buyer set forth in this Agreement shall have been true and correct in all material respects as of the date of this Agreement and shall be true and correct in all material respects on the Closing Date (or such other date(s) as specified, to the extent any representation or warranty speaks as of a specific date) with the same effect as though all such representations and warranties had been made on and as of such date; *provided, however*, that each of the representations and warranties in this Agreement that contains an express materiality qualification, shall have been true and accurate in all respects as of the date of this Agreement, and shall be true and accurate in all respects on the Closing Date (or such other date as specified) as if then made.

- (b) <u>Covenants</u>. Each and all of the covenants and agreements of Buyer to be performed or complied with at or prior to the Closing pursuant to this Agreement shall have been duly performed or complied with in all material respects by Buyer; *provided, however*, that each of the covenants in this Agreement that contains an express materiality qualification, shall have been performed or complied with in all respects at or prior to the Closing.
- (c) No Proceedings or Prohibitions. At the time of the Closing, there shall not be any litigation, investigation, inquiry or proceeding pending or threatened in or by any court or agency of any government or by any third party, that in the reasonable judgment of the executive officers of Seller, with the advice of counsel, presents a bona fide claim to restrain, enjoin or prohibit consummation of the Acquisition or that might result in rescission in connection with such transactions. No preliminary or permanent injunction or other order by any federal or state court that prevents the consummation of the Acquisition shall have been issued and shall remain in effect.
- (d) <u>Closing Documents</u>. Buyer shall have tendered for delivery all of the certificates, documents and other items set forth in <u>Section 2.8</u>.
- 8.2 <u>Conditions to Buyer's Obligations</u>. The obligation of Buyer to consummate the Acquisition is subject to the satisfaction, or the waiver in writing by Buyer to the extent permitted by applicable law, of the following conditions at or prior to the Closing:
- (a) <u>Representations and Warranties</u>. The representations and warranties of Seller set forth in this Agreement shall have been true and correct in all material respects as of the date of this Agreement and shall be true and correct in all material respects on the Closing Date (or such other date(s) as specified, to the extent any representation or warranty speaks as of a specific date) with the same effect as though all such representations and warranties had been made on and as of such date; *provided, however*, that each of the representations and warranties in this Agreement that contains an express materiality qualification, shall have been true and accurate in all respects as of the date of this Agreement, and shall be true and accurate in all respects on the Closing Date (or such other date as specified) as if then made.

- (b) <u>Covenants</u>. Each and all of the covenants and agreements of Seller to be performed or complied with at or prior to the Closing pursuant to this Agreement shall have been duly performed or complied with in all material respects by Seller; *provided, however*, that each of the covenants in this Agreement that contains an express materiality qualification, shall have been performed or complied with in all respects at or prior to the Closing.
- (c) No Proceedings or Prohibitions. At the time of the Closing, there shall not be any litigation, investigation, inquiry or proceeding pending or threatened in or by any court or agency of any government or by any third party that in the reasonable judgment of the executive officers of Buyer, with the advice of counsel, presents a bona fide claim to restrain, enjoin or prohibit consummation of the Acquisition or that might result in rescission in connection with such transactions. No preliminary or permanent injunction or other order by any federal or state court that prevents the consummation of the Acquisition shall have been issued and shall remain in effect.
- (d) No Material Adverse Change. There shall have been no material adverse change in the Acquired Assets between the date of this Agreement and the Closing Date (other than, with respect to the Loans, collection and work-out procedures undertaken in all material respects in accordance with Seller's normal and customary practices relating thereto and any related issues previously disclosed), provided, however, that a material adverse change for purposes of this Section 8.2(d) shall not be deemed to include (i) the effects of any change attributable to or resulting from changes in political, economic or market conditions, laws, regulations or accounting guidelines applicable to comparable depository institutions generally or in general levels of interest rates, (ii) changes or proposed changes after the date hereof in applicable law, (iii) the impact of the disclosure of the transactions contemplated hereby, including, but not limited to, relationships with the Acquired Customers or Acquired Employees, (iv) changes or proposed changes after the date hereof in GAAP or authoritative interpretation thereof, (v) actions taken by a party pursuant to the terms of this Agreement or with the written consent of the other party hereto, (vi) the issuance or compliance with any directive or order of any governmental authority, or (vii) any outbreak, escalation or worsening of hostilities, declared or undeclared acts of war, sabotage, military action or terrorism, or civil unrest.

- (e) <u>Closing Documents</u>. Seller shall have tendered for delivery all of the certificates, documents and other items set forth in Section 2.7.
- 8.3 <u>Waivers of Conditions Precedent</u>. The conditions specified in <u>Sections 8.1</u> and <u>8.2</u> herein shall be deemed satisfied or, to the extent not satisfied, waived if the Closing occurs, unless such failure of satisfaction is reserved in a writing executed by the waiving party at or prior to the Closing. Notwithstanding anything contained herein to the contrary, the parties acknowledge and agree that the conditions in <u>Section 8.1(a)</u> and <u>Section 8.2(a)</u> may not be waived.

#### Article IX Certain Transitional Matters

- Notices. On such date as the parties may agree, or in the absence of such agreement on the date that is the earlier of (a) thirty (30) days prior to the Closing Date of the Second Closing or the (b) minimum number of days prior to the Closing Date of the Second Closing required by any regulatory authority, Seller and Buyer shall jointly notify the Acquired Customers of the pending transfer of the Loans to Buyer. Such notice will be in a form acceptable to both parties and in compliance with all applicable laws and regulations. The cost of preparing and mailing such notice shall be borne by Buyer; provided that Seller shall provide the names and address of such holders at Seller's cost. Seller will cooperate with Buyer in providing such other notices to the Acquired Customers as Buyer may reasonably request. In addition, Buyer may, after the date hereof, at its own expense, communicate with and deliver information, brochures, press releases and other communications to the Acquired Customers concerning the Acquisition; *provided, however*, that all such communications shall be subject to reasonable approval by Seller and Seller, if so requested by Buyer, shall, on behalf of Buyer and at Buyer's sole expense, furnish such information and communications to customers in a reasonable manner.
- 9.2 <u>Loans</u>. Within ten (10) Business Days following the Closing Date of the Second Closing, Buyer shall prepare and transmit at Buyer's expense to each of the Obligors on Loans a notice to the effect that the Loan has been transferred and directing that payment be made to Buyer at the address specified by Buyer, with Buyer's name as payee on any checks or other instruments used to make payments, and, with respect to such Loan on which a payment notice or coupon book has been issued, to issue a new notice or coupon book reflecting the name and an address of Buyer as the person to whom and place at which payments are to be made.
- 9.3 <u>Interest Reporting.</u> Seller shall prepare and provide to Acquired Customers such reports as may be required by regulatory authorities and the Internal Revenue Service for all transactions occurring on or before the Closing Date. Buyer shall prepare and provide to the Acquired Customers such reports as may be required by any applicable regulatory authorities and the Internal Revenue Service for all transactions occurring after the Closing Date. Any such reports shall be made, as applicable, to the Acquired Customers, to the Regulatory and Authorities and to the Internal Revenue Service.

#### 9.4 Data Conversion.

(a) As soon as practicable following the date of this Agreement, Seller shall provide Buyer with applicable product functions and specifications relating to the data processing support required for the Loans to be acquired hereunder. As soon as practicable following the date of this Agreement, Seller shall engage its primary data processing provider to provide to Buyer, at Buyer's expense, file formats relating to the Loans and up to two (2) sets of test tapes related to the Loans in Seller's standard format, including any document images. Buyer shall review and analyze such materials, including the file formats and test tapes, and shall advise Seller in writing of any defects or concerns relating thereto not later than thirty (30) Business Days following receipt thereof. Seller shall reasonably cooperate with Buyer, at Buyer's expense, including any expense incurred with Seller's data processor, in obtaining test tapes or other information concerning the Loans in formats other than Seller's standard format.

27

(b) With respect to any other data, including any document images, or information relating to the Business, Acquired Employees or Acquired Assets, Seller shall cooperate with Buyer in connection with transition planning and preparing for the conversion of such data or information onto Buyer's platforms.

- (c) In furtherance of the foregoing, promptly after the date hereof, Seller and Buyer shall each appoint representatives as their respective transition representatives, who shall be responsible for the development and implementation of a transition plan describing milestones, responsible parties, and timelines for the integration, data conversion and migration planning of the Loans, Business and other Acquired Assets to Buyer, effective as of the First Closing (with respect to the Existing Loans and related Loan Agreements and Records), each Subsequent Loan Purchase Date (with respect to applicable Subsequent Loans acquired on such date) or as of the Second Closing otherwise, which plan shall be completed prior to the applicable Closing.
- 9.5 <u>Transition Services Agreement</u>. Buyer and Seller shall execute and deliver at Closing a Transition Services Agreement in substantially the form of <u>Exhibit E</u> pursuant to which employees of Seller will provide services to Buyer for a period of time following the Second Closing as described therein.

### Article X Additional Covenants

- 10.1 <u>Further Assurances</u>. From and after the date hereof, each party hereto agrees to execute and deliver such instruments and to take such other actions as the other party hereto may reasonably request in order to carry out and implement this Agreement.
- 10.2 <u>Information After Closing</u>. For a period of six years following the Second Closing, Buyer shall retain and preserve the Records of the Business that were acquired by Buyer hereunder and in existence prior to the Second Closing. Upon written request of Seller to Buyer, Buyer shall provide Seller, at Seller's expense, with reasonable access to, or copies of, such information and Records relating to the Business to permit Seller or its Affiliates to comply with or contest any applicable legal, Tax, banking, accounting or regulatory policies or requirements, or any legal or regulatory proceeding thereunder or requests related to customer relationships at the Business, in each case of this <u>Section 10.2</u> to the extent related to Seller's ownership or operation of the Business prior to the Second Closing.
- 10.3 <u>Timing of Closing</u>. Buyer and Seller agree to cooperate and use commercially reasonable efforts to consummate each of the First Closing, Subsequent Loan Purchase Date and Second Closing at the dates and times contemplated hereunder.
- 10.4 <u>Termination of Intercompany Arrangements</u>. Prior to the Second Closing, Seller shall take, or cause to be taken, all such actions reasonably necessary so that all outstanding payables, receivables, transactions, agreements or arrangements between Seller or its Affiliates, on the one hand, and the Business, on the other hand, are terminated, settled or cancelled with no liability or obligation on behalf of the Business from and after the Second Closing.

28

#### 10.5 Tax Matters.

- (a) <u>Cooperation</u>. Buyer and Seller agree to furnish or cause to be furnished to each other, upon request, as promptly as practicable, such information and assistance relating to the Business, Acquired Assets and the Assumed Liabilities (including reasonable access to books and records) as is reasonably necessary for the filing of all Tax Returns, interest reporting described in <u>Section 9.3</u> herein, the making of any election relating to Taxes, the preparation for any audit by any taxing authority, and the prosecution or defense of any claim, suit or proceeding relating to any Tax. Buyer and Seller agree to cooperate with each other in the conduct of any audit or other proceeding relating to Taxes involving the Business, the Acquired Assets or the Assumed Liabilities. The party requesting such cooperation shall bear the reasonable out-of-pocket costs of the other party.
- (b) <u>Transfer Taxes</u>. All sales, use, transfer, stamp duty or similar Taxes incurred in connection with the transaction contemplated in this Agreement shall be borne 50% by Seller and 50% by Buyer. The party legally required to do so will prepare and file all necessary Tax Returns and other documentation with respect to such Taxes and, if required by applicable law, the other party will (and will cause its Affiliates to) join in the execution of any such Tax Returns and other documentation. The parties shall cooperate in connection with the foregoing obligations.
- (c) <u>Straddle Period</u>. In the case of any Straddle Period, the amount of any Taxes imposed on a periodic basis with respect to the Existing Loans (or other assets acquired in the First Closing) and Acquired Assets (other than any Tax described in <u>Section 10.5(b)</u>, above) that relate to a Pre-Closing Tax Period will be deemed to be the amount of such Tax for the entire Taxable period

multiplied by a fraction, the numerator of which is the number of days in the Taxable period ending on the applicable Closing Date and the denominator of which is the number of days in such Straddle Period. Seller shall be liable for the amount of such Taxes that are attributable to the portion of the applicable Straddle Period ending on and including the applicable Closing Date, and Buyer shall be liable for the amount of such Taxes that are attributable to the remaining portion of the applicable Straddle Period. If any such Taxes are paid by Seller or Buyer on behalf of the other, Seller and Buyer shall cooperate to promptly reimburse the other for any such Taxes based on their respective liability for such Taxes as determined pursuant to this Section 10.5(c). Any refunds of such Taxes with respect to a Straddle Period shall be apportioned between Seller and Buyer in a similar manner. Notwithstanding the forgoing, items attributable to any action taken by Buyer after the Closing Date that is not in the ordinary course of business will not be attributable to a Pre-Closing Tax Period.

#### Article XI Termination

- 11.1 Termination. This Agreement may be terminated at any time prior to the Closing Date:
  - (a) By mutual written consent of Buyer and Seller;

29

- (b) By Buyer if: (i) any of the conditions in <u>Section 8.2</u> of this Agreement has not been satisfied on or before the Termination Date, or if satisfaction of such a condition is or becomes impossible (other than through the failure of Buyer to comply with its obligations under this Agreement); and (ii) Buyer has not waived such condition on or before the Closing Date;
- (c) By Seller if: (i) any of the conditions in <u>Section 8.1</u> of this Agreement has not been satisfied on or before the Termination Date, or if satisfaction of such a condition is or becomes impossible (other than through the failure of Seller to comply with its obligations under this Agreement); and (ii) Seller has not waived such condition on or before the Closing Date; or
- (d) By either party after the Termination Date, in the event the Second Closing has not been consummated by such date (provided that the terminating party is not then in material breach of any representation, warranty, covenant or other agreement contained herein).
- 11.2 <u>Effect of Termination</u>. If this Agreement is terminated pursuant to <u>Section 11.1</u>, this Agreement shall forthwith become void, there shall be no liability under this Agreement on the part of Buyer, Seller or any of their respective Affiliates, and all rights and obligations of each party hereto shall cease; *provided*, *however*, that nothing herein shall release any party from liability for a breach of any of its representations or warranties or a willful breach of any of its covenants or agreements set forth in this Agreement. The provisions contained in <u>Section 13.6</u> shall survive the termination of this Agreement.

### Article XII Indemnification

- Indemnification by Seller. Following the Closing Date, Seller agrees to indemnify Buyer against, and hold Buyer, its Affiliates, and their respective directors, officers, employees and agents (collectively, the "Buyer Indemnified Parties") harmless from any claim, loss, damage, penalty, fine, costs, or expenses (including reasonable attorneys' fees and costs and expenses of litigation) (collectively, "Losses") directly or indirectly relating to or arising out of: (a) any breach of any representation or warranty made by Seller herein, (b) any breach of any of the agreements or covenants of Seller contained herein, (c) any Excluded Asset or Excluded Liability, or (d) the failure of any Loan to be solicited, originated, administered or serviced in accordance with applicable laws, regulations and rules of applicable jurisdictions, including any inability to enforce the obligations under such Loan or collect upon or get a security interest with respect to any collateral underlying such Loan solely due to such failure.
- 12.2 <u>Indemnification by Buyer</u>. Following the Closing Date, Buyer agrees to indemnify Seller against, and hold Seller, its Affiliates, and their respective directors, officers, employees and agents (collectively, the "Seller Indemnified Parties") harmless from any Losses directly or indirectly relating to or arising out of: (a) any breach of any representation or warranty made by Buyer herein, (b) any breach of any of the agreements or covenants of Buyer contained herein, or (c) any Assumed Liability, including any failure by Buyer to fulfill the liabilities and obligations associated with the Existing Loans and Acquired Assets and assumed by Buyer in the Acquisition as expressly set forth herein.

- 12.3 Third-Party Claims. In the event any action, suit or other proceeding shall be commenced or a claim shall be made against either a Buyer Indemnified Party or a Seller Indemnified Party as to which such party may assert a right of indemnification under this Article XII (each, an "Indemnified Party"), such Indemnified Party shall give prompt written notice thereof to each person obligated to indemnify the Indemnified Party under this Article XII (each, an "Indemnifying Party"), and such Indemnifying Party shall be entitled to control the defense thereof, including employment, at such Indemnifying Party's expense, of counsel reasonably satisfactory to the Indemnified Party; provided, however, that an Indemnifying Party shall not have the right to assume and control the defense of any criminal claim or action, regulatory claim or action brought by any Governmental Entity or other claim or action that seeks material nonmonetary relief, provided, further, that if the Indemnifying Party controls the defense of any such action, suit or other proceeding (i) it will be conclusively established for purposes of Sections 12.1 or 12.2, as applicable, that the claims made therein are within the scope of and subject to indemnification hereunder, (ii) no compromise or settlement of such claims may be effected by the Indemnifying Party without the Indemnified Party's consent unless (1) the sole relief provided is monetary damages that are paid in full by the Indemnifying Party, (2) there is no finding or admission of any violation of law or regulation or wrongdoing by any Indemnified Party or its Affiliates (or the Business if the Indemnified Party is a Buyer Indemnified Party) and (3) the Indemnified Party and its Affiliates is released from any further liabilities with respect to such matter, and (iii) the Indemnified Party will have no liability with respect to any compromise or settlement of such claims effected without its consent; provided, further, that the Indemnified Party may employ, at the Indemnifying Party's expense, its own counsel if its interests are, in its reasonable opinion, in conflict with those of the Indemnifying Party or are not being adequately represented. In the event the Indemnifying Party does not elect to control the defense of any claim within ten (10) Business Days of the notice thereof from the Indemnified Party, the Indemnified Party may contest, or upon prior approval of the Indemnifying Party (not to be unreasonably withheld), settle any claim or liability that, if established, would be subject to indemnification hereunder and, in such event, all reasonable legal fees, disbursements and other costs and expenses of such contest or settlement shall also be an item of indemnification hereunder.
- 12.4 Survival; Limitations. Except as waived in accordance with the terms of this Agreement, any provision of this Agreement that imposes an obligation or restriction, or confers a right or benefit, the observance, performance, or exercise of which may or must occur after the Closing Date, shall survive the Closing Date. Notwithstanding the foregoing, (i) the representations and warranties contained in this Agreement shall survive for a period of twelve (12) months following the Closing Date (other than the Fundamental Representations and the representations set forth in Section 6.1(e) which shall survive until the thirty (30) days following expiration of the statute of limitations) and (ii) except with respect to breaches of Fundamental Representations, (A) neither Seller nor Buyer shall have any obligation to indemnify an Indemnified Party pursuant to Sections 12.1(a) and 12.1(b), on the one hand, or 12.2(a) and 12.2(b), on the other hand, as applicable, until the aggregate amount of all Losses indemnifiable under Sections 12.1(a) and 12.1(b), on the one hand, or Sections 12.2(a) and 12.2(b), on other hand, as applicable, exceeds \$500,000, after which Seller or Buyer (as the case may be) shall indemnify the Indemnified Party for all Losses of in excess of such amount and (B) neither Seller nor Buyer shall have any indemnity obligation to an Indemnified Party for aggregate Losses indemnifiable under Sections 12.1(a) and 12.1(b), on the one hand, or 12.2(a) and 12.2(b), on other hand, as applicable, in excess of \$1,500,000 (the "Cap"); provided, however, with respect to breaches of the Fundamental Representations, the Cap shall be the Premium; provided further that any indemnification pursuant to Section 12.1(a) shall not be in duplication of indemnification pursuant to Section 12.1(d); provided further the Cap and the \$500,000 deductible set forth above shall not apply to any breach of Section 4.7. For the avoidance of doubt, the Cap shall not apply to any indemnification pursuant to clauses Sections 12.1(c), 12.1(d) or 12.2(c). The right to indemnification, reimbursement or other remedy based upon such representations, warranties, covenants and obligations (other than those explicitly waived in accordance with the terms of this Agreement) will not be affected by any investigation conducted with respect to, or any knowledge acquired (or capable of being acquired) at any time, whether before or after the execution and delivery of this Agreement or the Closing Date, with respect to the accuracy or inaccuracy of or compliance with any such representation, warranty, covenant or obligation.

12.5 <u>Remedies.</u> After the Closing, except as expressly provided otherwise in this Agreement, and except in the case of fraud, this Article XII shall provide the exclusive remedy for any breach of any representation or warranty set forth in this Agreement or otherwise arising out of this Agreement or the transaction contemplated hereby; provided that the foregoing shall not prevent a party from obtaining specific performance, injunctive relief or any other available non-monetary equitable remedy.

### Article XIII Miscellaneous Provisions

- 13.1 <u>Entire Agreement</u>. This Agreement, the Exhibits and Schedules hereto and the instruments, agreements, certificates and documents contemplated hereby supersede all other prior or contemporaneous understandings, commitments, representations, negotiations, discussions and agreements, whether oral or written or express or implied, between the parties hereto relating to the matters contemplated hereby, and constitute the entire agreement between the parties hereto relating to the Acquisition.
- 13.2 <u>Binding Effect; Assignment</u>. This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns; *provided*, *however*, that neither party may assign this Agreement without the prior written consent of the other party.
- 13.3 <u>Amendment and Modification</u>. The parties hereto may amend, modify or supplement this Agreement only by an agreement in writing executed by Seller and Buyer.
- 13.4 <u>Waiver or Extension</u>. Either party hereto may by an instrument in writing waive the performance by the other of any of the covenants or agreements to be performed by such other party under this Agreement; *provided, however*, that neither party may waive the requirement for obtaining any applicable regulatory approvals required hereunder. The failure of either party hereto at any time to insist upon the strict performance of any covenant, agreement or provision of this Agreement shall not be construed as a waiver or relinquishment of the right to insist upon strict performance of such covenant, agreement or provision at a future time. The waiver by any party hereto of a breach of or noncompliance with any provision of this Agreement shall not operate or be construed as a continuing waiver or a waiver of any other or subsequent breach or noncompliance hereunder.
- 13.5 <u>Payment of Expenses</u>. Except as otherwise expressly provided in this Agreement, each party hereto shall bear and pay all costs and expenses incurred by it or on its behalf in connection with this Agreement and the Acquisition.

32

- Confidentiality; Press Releases. Each of Buyer and Seller shall, and shall cause its directors, officers, employees, agents and representatives to, hold in strict confidence and not disclose to any other person or entity without the prior written consent of the other the terms of this Agreement, except as may be otherwise provided herein or as required by law. Buyer and Seller agree that neither shall issue any news or press release nor provide information to any reporter or the media regarding this Agreement or the Acquisition, except as is required by applicable law, without obtaining the prior approval of the other party. Notwithstanding the foregoing, Buyer acknowledges and agrees that Seller may announce the transactions contemplated by this Agreement in its quarterly earnings release or otherwise in Seller's public filings pursuant to the Securities Exchange Act of 1934. Upon the termination of this Agreement, Buyer and Seller shall, and shall cause each of their respective Affiliates and representatives to, promptly return all documents and work papers containing, and all copies of, any confidential information (so required to be treated as confidential) received from or on behalf of the other party in connection with the Acquisition. The covenants of Buyer and Seller contained in this Section 13.6 shall survive any termination of this Agreement.
- Notices. All notices, requests and other communications hereunder shall be in writing (which shall include facsimile communication) and shall be deemed to have been duly given (a) when delivered by hand and receipted for, (b) five (5) days after having been sent by certified United States Mail, return receipt requested, first class postage pre-paid, (c) when delivered by receipted overnight delivery service or (d) when delivered by facsimile or email transmission if such fax or email is confirmed, immediately thereafter by also mailing a copy of such notice, request or other communication by certified United States Mail, return receipt requested, first class postage pre-paid, in each case as follows:

If to Seller to: Primis Bank

6830 Old Dominion Drive McLean, VA 22101

Attention: Matthew A. Switzer

E-mail: matthew.switzer@primisbank.com

with a copy to: Alston & Bird LLP

1201 West Peachtree Street

Atlanta, GA 30309 Attention: David S. Park E-mail: david.park@alston.com

If to Buyer to: EverBank, N.A.

Attention: Mark Baum

Email: mark.baum@everbank.com

with a copy to:

Simpson Thacher & Bartlett LLP

425 Lexington Avenue New York, New York 10017 Attention: Ravi Purushotham Email: RPurushotham@stblaw.com

33

or such substituted address or person as either party has given to the other in writing.

- 13.8 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts, each of which shall be an original, but such counterparts shall together constitute one and the same agreement.
- 13.9 <u>Governing Law.</u> This Agreement shall be governed by and construed in accordance with the internal laws of the State of Delaware, without regard to any choice or conflict of law provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of the laws of any jurisdiction other than the State of Delaware.
- 13.10 Severability. In case any one or more of the provisions (or any portion thereof) contained herein shall, for any reason, be held to be invalid, illegal or unenforceable in any respect, such invalidity, illegality or unenforceability shall not affect any other provision of this Agreement, but this Agreement shall be construed as if such invalid, illegal or unenforceable provision or provisions (or portion thereof) had never been contained herein.
- 13.11 <u>No Third-Party Rights</u>. Nothing in this Agreement, expressed or implied, is intended to confer upon any person or entity, other than the parties hereto, or their respective permitted successors and assigns, any rights, remedies, obligations or liabilities under or by reason of this Agreement.
- 13.12 <u>Construction</u>. This Agreement is the product of negotiation by both parties hereto and shall be deemed to have been drafted by both parties hereto. This Agreement shall be construed in accordance with the fair meaning of its provisions and its language shall not be strictly construed against, nor shall ambiguities be resolved against, either party.
- 13.13 <u>Schedules and Exhibits.</u> The Schedules and Exhibits attached hereto are incorporated into and made a part of this Agreement.

[Signature Page Follows.]

34

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed by their respective officers thereunto duly authorized, all as of the date first above written.

#### **SELLER:**

#### PRIMIS BANK

By: /s/ Matthew Switzer

Name: Matthew Switzer Title: Chief Financial Officer

#### **BUYER:**

#### **EVERBANK, NATIONAL ASSOCIATION**

By: /s/ Greg Seibly

Name: Greg Seibly

Title: Chief Executive Officer

[Signature Page to Purchase and Assumption Agreement]

### PRIMIS FINANCIAL CORP. UNAUDITED PRO FORMA CONDENSED CONSOLIDATED FINANCIAL INFORMATION

On October 31, 2024, Primis Bank (the "Bank"), a wholly-owned subsidiary of Primis Financial Corp. (the "Company"), completed the first closing of its previously announced sale of its Life Premium Finance divison to EverBank, N.A. ("EverBank"). The second closing and full transfer of the business, including employees, is scheduled for January 31, 2025.

The unaudited pro forma condensed consolidated financial information is based on our historical financial statements, and gives effect to the sale of the division in accordance with Article 11 of Regulation S-X. The unaudited pro forma condensed consolidated statements of income for the nine months ended September 30, 2024 and the year ended December 31, 2023 give effect to the sale of the division as if it had occurred on January 1, 2023, while the unaudited pro forma condensed consolidated balance sheet as of September 30, 2024 gives effect to the sale as if it had occurred on that day.

The unaudited pro forma condensed consolidated financial statements (i) do not necessarily reflect what the Company's financial condition or results of operations would have been had the sale of the division occurred on the dates indicated, (ii) are based upon available information and assumptions that management considers to be reasonable to give effect, on a pro forma basis, to the sale of the division and (iii) are intended for informational purposes only. In addition, the unaudited pro forma condensed consolidated financial statements should not be considered to be indicative of our future consolidated financial performance or results of operations and should be read in conjunction with our audited consolidated financial statements, and notes thereto, included in our Annual Report on Form 10-K for the year ended December 31, 2023 and in our Form 10-Q for the three and nine-months ended September 30, 2024.

1

### PRIMIS FINANCIAL CORP. UNAUDITED PRO FORMA CONDENSED CONSOLIDATED BALANCE SHEET

	<b>September 30, 2024</b>					
	Historical		Adjustments		Pı	o Forma
ASSETS						
Cash and cash equivalents:						
Cash and due from financial institutions	\$	10,854	\$	373,916(a)	\$	384,770
Interest-bearing deposits in other financial institutions		66,420		, ,		66,420
Total cash and cash equivalents		77,274		373,916		451,190
•						
Securities available-for-sale, at fair value (amortized cost of \$264,151 and \$255,891, respectively)		242,543				242,543
						_
Securities held-to-maturity, at amortized cost (fair value of \$9,162 and \$10,839, respectively)		9,766				9,766
Loans held for sale, at fair value		96,897				96,897
Loans held for sale, at lower of cost or market		361,825		(361,825)(b)		0
Total loans held for sale		458,722		(361,825)		96,897
Loans held for investment, collateralizing secured borrowings		17,591				17,591
Loans held for investment		2,956,132				2,956,132
Less: allowance for credit losses		(51,132)		435(c)		(50,697)
Net loans		2,922,591		435		2,923,026

	_		_		_	
Stock in Federal Reserve Bank (FRB) and Federal Home Loan Bank (FHLB)		20,875				20,875
Bank premises and equipment, net		19,668				19,668
Assets held for sale		9,864		(5,966)(d)		3,898
Operating lease right-of-use assets		10,465		(3,700)(u)		10,465
Cloud computing arrangement assets, net		8,460				8,460
Goodwill		93,459				93,459
Intangible assets, net		985				985
Bank-owned life insurance		66,750				66,750
Deferred tax assets, net		25,582				25,582
Consumer Program derivative asset		7,146				7,146
Other assets		50,197				50,197
Total assets	\$	4,024,347	Φ	6,560	\$	4,030,907
Total assets	Ф	4,024,347	\$	0,300	Ф	4,030,907
LIABILITIES AND STOCKHOLDERS' EQUITY						
Noninterest-bearing demand deposits	\$	421,231			\$	421,231
Interest-bearing deposits:	Ψ	1,_01			Ψ	1,_01
NOW accounts		748,833				748,833
Money market accounts		835,099				835,099
Savings accounts		873,810				873,810
Time deposits		427,458				427,458
Total interest-bearing deposits	_	2,885,200			_	2,885,200
Total deposits	_	3,306,431				3,306,431
Total deposits	_	3,300,431			_	3,300,431
Securities sold under agreements to repurchase		3,677				3,677
Secured borrowings		17,495				17,495
FHLB advances		165,000				165,000
Junior subordinated debt		9,867				9,867
Senior subordinated notes		85,941				85,941
Operating lease liabilities		11,704				11,704
Other liabilities		27,169		2,563(e)		29,732
Total liabilities	_	3,627,284	_	2,563		3,629,847
		-,,				-,,
Stockholders' equity:						
Preferred stock, \$0.01 par value. Authorized 5,000,000 shares; no shares issued						
and outstanding		_				_
Common stock, \$0.01 par value. Authorized 45,000,000 shares; 24,722,734 and						
24,693,172 shares issued and outstanding at September 30, 2024 and		247				247
December 31, 2023, respectively						
Additional paid in capital		314,066				314,066
Retained earnings		83,854		3,997(f)		87,851
Accumulated other comprehensive loss		(17,145)		-,,(-)		(17,145)
Total Primis stockholders' equity		381,022	_	3,997		385,019
Noncontrolling interests		16,041		5,771		16,041
Total stockholders' equity		397,063		2 007		401,060
Total liabilities and stockholders' equity	¢.		¢	3,997	¢.	
Total natimites and stockholders equity	\$	4,024,347	\$	6,560	\$	4,030,907

<sup>(</sup>a) Reflects the receipt of cash consideration at the closing of the transaction.

<sup>(</sup>b) Reflects the loan balances being sold net of unearned income.

<sup>(</sup>c) Reflects the recovery of allowance for credit losses on sold loans.

<sup>(</sup>d) Reflects the payment of accrued interest.

<sup>(</sup>e) Reflects accruals for transaction costs and income tax expense recorded at closing.

<sup>(</sup>f) Reflects the gain recorded at the closing of the transaction, net of transaction costs, recognition of unearned income, recovery of allowance for credit losses and income tax expense.

### PRIMIS FINANCIAL CORP. UNAUDITED PRO FORMA CONDENSED CONSOLIDATED STATEMENT OF INCOME

	For the Nine Months Ended September 30, 2024					For the Year Ended December 31, 2023						
	Historical		Adjustments		Pro Forma		Historical		Adjustments		Pro Forma	
Interest and dividend income:	Historica		<u> </u>		<u>o i oi iii.</u>		astor retar	114	justinents		0 1 01 1114	
Interest and fees on loans	\$ 151,58	1 \$	(16,610)(g)	\$	134,971	\$	172,788	\$	(14,635)(g)	\$	158,153	
Interest and dividends on taxable securities	5,02	2	_		5,022		5,966		_		5,966	
Interest and dividends on tax exempt securities	29	7	_		297		407				407	
Interest and dividends on other earning assets	2,75	6	_		2,756		13,457		_		13,457	
Total interest and dividend income	159,65		(16,610)		143,046		192,618		(14,635)		177,983	
Interest expense:			( 1)1 1)		- )		- ,				, ,	
Interest on deposits	72,98	7	_		72,987		83,690				83,690	
Interest on other borrowings	8,52		(10,527)(g)		(2,003)		10,217		(10,356)(g)		(139)	
Total interest expense	81,51		(10,527)	_	70,984	_	93,907		(10,356)	_	83,551	
Net interest income	78,14		(6,083)		72,062		98,711	_	(4,279)		94,432	
Provision for credit losses	17,13		(277)(g)		16,861		32,540		(4,277) $(60)(g)$		32,480	
Net interest income after provision for	17,13		(2//)(g)		10,001		32,340		(00)(g)		32,400	
credit losses	61,00	7	(5,806)		55,201		66,171		(4,219)		61,952	
Noninterest income:												
Account maintenance and deposit service fees	4,72	2	(5)(g)		4,717		5,733		1(g)		5,734	
Income from bank-owned life insurance	1,97		(3)(g)		1,975		2,021		1(g)		2,021	
Mortgage banking income	18,77		<del>_</del>		18,779		17,645		_		17,645	
Gain on other investments	39				393		184				184	
Consumer Program derivative gains (losses)	3,39				3,392		18,120				18,120	
Other noninterest income	1,18				1,180		1,547		_		1,547	
Gain on Disposal	1,10		4,500(h)		4,500				4,500(h)		4,500	
Total noninterest income	30,44	1	4,495		34,936		45,250		4,501		49,751	
Nonintorest expenses												
Noninterest expenses: Salaries and benefits	48,58	7	(494)(g)		48,093		58,765		(377)(g)		58,388	
Occupancy expenses	3,98		(494)(g)		3,988		6,239		(3//)(g)		6,239	
Furniture and equipment expenses	5,28		(10)(g)		5,278		6,381		(113)(g)		6,268	
Amortization of intangible assets	95		(10)(g)		952		1,269		(113)(g)		1,269	
Virginia franchise tax expense	1,89				1,894		3,395				3,395	
FDIC insurance assessment	1,74				1,744		2,929		_		2,929	
Data processing expense	7,13				7,130		9,545		(1)		9,544	
Marketing expense	1,40		(3)(g)		1,404		1,819		(1)(g)		1,818	
Telephone and communication expense	1,01		(3)(g)		1,014		1,507		—(g)		1,507	
Loss on bank premises and equipment and assets	-	_	—		_		476		_		476	
held for sale Professional fees	7.05	5	((7)()		7 100		1 (11		(-)		1 6 1 1	
	7,25	J	(67)(g)		7,188		4,641		—(g)		4,641	
Goodwill impairment Fraud losses	_	9	<del>-</del>		59		11,150 3,311		<del>-</del>		11,150	
Miscellaneous lending expenses	1,83		_		1,835		3,006				3,311 3,006	
Other operating expenses	7,12				7,069		8,167		(38)(g)		8,129	
	_		(54)(g)			_		_				
Total noninterest expenses	88,27		(631)		87,648		122,600		(529)		122,071	
Income (loss) before income taxes	3,16		(680)		2,489		(11,179)		811		(10,368)	
Income tax expense (benefit)	1,67		(143)(i)		1,536		(1,067)		170(i)		(897)	
Net income (loss)	1,49	U	(537)		953		(10,112)		641		(9,471)	
Net income attributable to noncontrolling interests	5,64	0	_		5,640		2,280		_		2,280	

Net income (loss) attributable to Primis' common stockholders	\$	7,130	\$	(537)	\$	6,593	\$	(7,832)	\$	641	\$	(7,191)
Other comprehensive income (loss):												
Unrealized loss on available-for-sale securities		5,863		_		5,863		5,250		_		5,250
Tax benefit		1,231				1,231		1,177				1,177
Other comprehensive loss		4,632				4,632		4,073				4,073
Comprehensive income (loss)	\$	11,762	\$	(537)	\$	11,225	\$	(3,759)	\$	641	\$	(3,118)
Earnings (loss) per share, basic	\$	0.29	\$	(0.02)	\$	0.27	\$	(0.32)	\$	0.03	\$	(0.29)
Earnings (loss) per share, diluted	\$	0.29	\$	(0.02)	\$	0.27	\$	(0.32)	\$	0.03	\$	(0.29)
											_	
Weighted average basic common shares	24	,683,556	24	,683,556	24	,683,556	24	4,638,609	24	4,638,609	24	1,638,609
Weighted average diluted common shares	24	,710,345	24	,710,345	24	,710,345	24	4,638,609	24	4,638,609	24	1,638,609

<sup>(</sup>g) Reflects the elimination of revenues and expenses associated with the division for the nine months ended September 30, 2024 and the year ended December 31, 2023.

### PRIMIS FINANCIAL CORP. NOTES TO UNAUDITED PRO FORMA CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

#### (1) Basis of Presentation

The unaudited pro forma condensed consolidated statements of income for the nine months ended September 30, 2024 and the year ended December 31, 2023 give effect to the sale of the division as if it had occurred on January 1, 2023, while the unaudited pro forma condensed consolidated balance sheet as of September 30, 2024 gives effect to the sale as if it had occurred on that day.

#### (2) Adjustments to the Unaudited Pro Forma Condensed Consolidated Financial Statements

The following is a summary of the transaction accounting adjustments reflected in the unaudited pro forma condensed consolidated financial statements:

- (a) Reflects the receipt of cash consideration at the closing of the transaction.
- (b) Reflects the loan balances being sold net of unearned income.
- (c) Reflects the recovery of allowance for credit losses on sold loans.
- (d) Reflects the payment of accrued interest.
- (e) Reflects accruals for transaction costs and income tax expense recorded at closing.
- (f) Reflects the gain recorded at the closing of the transaction, net of transaction costs, recognition of unearned income, recovery of allowance for credit losses and income tax expense.
- (g) Reflects the elimination of revenues and expenses associated with the division for the nine months ended September 30, 2024 and the year ended December 31, 2023.
- (h) Reflects the pre-tax gain recorded at the closing of the transaction, net of transaction costs.

<sup>(</sup>h) Reflects the pre-tax gain recorded at the closing of the transaction, net of transaction costs.

<sup>(</sup>i) Reflects the estimated tax impact of the transaction accounting adjustments.

4

(i) Reflects the estimated tax impact of the transaction accounting adjustments.

Cover Oct. 31, 2024

**Cover** [Abstract]

Document Type 8-K
Amendment Flag false

Document Period End DateOct. 31, 2024Entity File Number001-33037

Entity Registrant Name Primis Financial Corp.

Entity Central Index Key 0001325670 Entity Tax Identification Number 20-1417448

Entity Incorporation, State or Country Code VA

Entity Address, Address Line One 1676 International Drive, Suite 900

Entity Address, City or Town
Entity Address, State or Province
Entity Address, Postal Zip Code
City Area Code
Local Phone Number

Written Communications

McLean

McLean

703

22102

703

Local Phone Number

Written Communications false
Soliciting Material false
Pre-commencement Tender Offer false
Pre-commencement Issuer Tender Offer false

<u>Title of 12(b) Security</u> COMMON STOCK

Trading SymbolFRSTSecurity Exchange NameNASDAQ

Entity Emerging Growth Company false

```
[
Twending': "2.2",
Twending': "1.2",
Twendinger': [
*smooth in Twend': [
*smooth in Twend': [
*smooth in Twend': "Single / per in India
*disp': [
*smooth in Twend': [
*smooth i
                                                            The state of the s
                                                                      "labellich": |
"lamel": |
"fess-20241031_lab_ses1"
Temporal Control of the Control of t
                                                                                "Bandy",
"Saud"
"Facilities of the Control of the C
                                                                                          lr
"reportCount": 1,
"baseSed": "tm242763045_Sh.htm",
"first": true,
"unique": true
                                                            |
| "sag" |
| "ded lamendmentDmanrightom" |
| "ded lamendmentDmanrightom" |
| "ded layer" | "stringlendings" |
| "marriage" | "stringlendings" |
| "marriage" | "lamendmentDmanrightom" |
| "marriage" | "lamendmentDmanrightom" |
| "mathy //primishanh.com/enle/Cover" |
| "http://primishanh.com/enle/Cover" |
                                                                                          ];
"Tang": {
"Tang": Tangedomnia Encociption",
"Tangedomnia Tangedomnia Encociption",
"Tangedomniation": "Encociption of obseque contained within assended document."
                                                                                                              ineq": {
"error": {
"error": {
"ende": {
"inde": {
"inde
                                                                                          "Making and "Jack Cornellations"; [
"Making and "Jack Cornellations"; [
"Making and "Jack Cornellations", "Making Cornellation
                                                                                The disconnection of the property of the control of
                                                                                ]
]: DoromenhFienalPerindForum*: {
"skellype": "FiscalPerindForum*: {
"skellype": "FiscalPerindForum*: project / 2222";
"lesalmen": "DomenshiperindForum*;
Teraniment": "DomenshiperindForum*;
Tyrementation": {
"Aktys/PyrindshiperindForum*;
Tyrementation": {
"Aktys/Pyrind
                                                                                          THE CONTRACT OF THE PROPERTY O
```

```
[ang's] {

"serve";

["serve";

"discolor Tomment Quarterly Reports,

"dommentation": Thomas Eigensteen an operatorly report.
                     "Nature ("Parisabanha mend manerumen"

"Ange" | {
    "second" {
    "second" {
    "second" {
    "second" {
    "second" {
    "second" {
    "second {

                     [c]
[dagg*: [d]
[dagg*: ]

"angur'y gwannandarum.
Teorati's (
"Berna"s (
"Berna"s (
"Bala"s (
"Tala"s (
       ]

Thei Ensitybidiemanhidiemanlinei", [
"desi Ensitybidiemanhidiemanlinei", [
"desitype", "massanlineidlaringitsmilype",
"desitys", "blays, //blab.liner.gov/des/2023*,
"massanlineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidlarineidl
                                    ang". [
'morus'. [
'mole'. [
'Take'. Thirty Address, Address Line One",
'Take'. "Entity Address, Eddress Line One",
'downwartshine'. 'Eddress Line I much as Bits, Building Esse, Etreet Esse'
       ]:
"Meag": [
"Recul": [
"Recul": [
"Subd": "Entity Address, Address Lice Des",
"Associational Conference Lice Des",
"Associational Conference Lice 2 much as Street or Suite number"
                             long": {
"serve": {
"serve": {
"sele": {
"Likel": Entity Address, Address Live Them",
"Likel": "Entity Address, Address Live Them",
"decommodation": "Address Live I nuch as an Office Park"
              [;
"Rang": {
"marma": {
"mala": {
"label": "Estity Address, City or Town",
"documentation": "Mame of the City or Town",
                     [c
Tang": [
Tenum": [
Tenum": [
Tenum": [
Tidesummanation": TIDO INSE: John-1 country code."
              ing": {
"error": {
"end": {
"end": {
"end": {
"inde": {
"finity Address, Pastal Eig Code",
"documentation": "Code for the postal or sig code",
                             Long": (
"Section": (
"section": (
"selec": (
"label": "Entity Address, State or Furnises",
"documentation": "Ease of the state or province."
The state of the s
              )
ly
"month and" : [
"ad"
       ]
]-dei EntisyCentralIndemEny*: {
"minitype": "mentralIndemEny*: {
"minitype": "mentralIndemEny*: mentralIndemEny*
"mentr": "hitps://minit.men.que/dei/2021",
"incaliname": "MinityCentralIndemEny",
"premonanties": [
"hitps://minithame.com/enie/Cores*
                             insert | 1
**The second of the second of the
       ]

[India Tenting Commonwhile and Contenting of a leading of the Contenting of the C
```

```
iner" (
"Tenned |-
"Te
      ".
"desi EnsisyWildlunders": [
"desi EnsisyWildlunders": [
"desillyym": "filedunders!temTyym",
"nessi": "hitps://shri.mer.gov/des/2023",
"finesilmens": "AntityWildlunders",
Tyrementation": [
"hitps://grinchland.com/sole/Coree"
             Name / Na
                    "del EntisyPublicFloat": {

"del EntisyPublicFloat": {

"del EntisyPublicFloat": {

"del Entisy //del.ce..gev/del/2023",

Torsel": "this y //del.ce..gev/del/2023",

Torselmose", "fortisyPublicFloat",

Torselmose",

Torselmose',

Torselmose'
      Format's (
**mark's (

             "Names (Princetable Association) educations "Services" |
"Agent" |
"Agent " |
                    lang's (
**Teachers's (
**Teachers's
             "Targ"; [
"Second"; [
"Manager To Manager To
      "Manager of the Committee of the Committ
                           | Tabemaion's | |
"del Labemaion's |
"del Labemaion's |
"delsigne": "marmalisedStringStemSype",
"mars!". "https://del.nor.gov/del/2023",
"leanings": "falmenion".
"presentation's |
"batts://primichant.com/sole/Cover"
             Jang": {
    "sule": {
    "sule": {
    "sule": {
    "sule": {
    "sule": {
    "decumentation": "Local Phone Number for entity."
    "decumentation": "Local phone number for entity."
                    [langs.]

Tamours [

Tamours |

T
                    "Native proposedaments"

"State of the Company of t
```

```
]

"dei_FreCommencement Tender of Err" : {

"dei_Stype": "boolses Innergye";

"more!": "boby://dei.er.gov/dei/2021";

Tons Inner": "PerCommencement Tender of Err";

Tymenchastion": {

"boby://gainthanh.com/male/Crows"
                                                                         Teach red": (
                                                             ]

[January 12] State of the st
                                                                         )
la
Taobh seilTo (
Taobh
                                    Tangar (
Tangar )

Tangar (
Tang
                                                                                      ];
Tangf' {
    Tenuf' {
    Tenuf' {
    Tenuf' {
    Tenuf' {
    Tenuf' {
    Tenuf' {
    Time of {
    Time o
                                                 In
Testin and To I
],
"sil", (
"sil", Tabings//www.aksl.neg/2003/solm/pemmoha
"sil", "Sill",
"sill", "Sill",
"Sill",
"Sill", "Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sill",
"Sil
            "Glanding of The "Granding of The "Granding of The Granding of
            "State of the Control of the Control
            "Schematical Conf."
"Self-1 | Self-2 | 
            "gate | "aby / wee.wki.seg/2011/swis/presentationDef",
"sain" | "Early ("EEE",
"East" ("EEE"),
"East" ("Early East )
"East" ("Early East )
"East )
"East ("Early East )
"East ("East )
"East ("East
                        "Submension": "6"
'[202": ["mois": "bhigs //www.sholoog/2005/solo/yemmensholoofie",
'Toular': "Toular': "Editor'
'Toular': "Inchange Joh",
'Toular': "Inchange Joh",
'Toular': "20";
'Submension': "20";
'Toular': "20";
                        ];
"Sill" [ "Nobys//our.skel.org/2011/sole/presentationSelf",
Table" "Things//our.skel.org/2011/sole/presentationSelf",
Thing' "Thing 10",
Thing' "Thing 10",
Thing 10", "Thing 10",
Thing 
            "Sanking", "320"
[*217", [**]
"sala", "haky //www.wki.weg/2003/sala/pweasekahindhel",
"sala", "haky //www.kii.weg/2003/sala/pweasekahindhel",
"Salaa", "Slow 13-0",
"Salaa", "Slow 13-0",
"Salaawina", "Slow 13-0",
"Salaawina", "s"
[**]
            "Makementaris" 1782
"Selection of Telegraphy (Makementaris 1782)
"Selection of Telegraphy (Makementaris 1782) (Makementaris 1782)
"Selection of Telegraphy (
            "Sensence: "map.
"Street, "The property of the sense of t
```

Supply 6 889 AMERICAN STREET STREET STREET